## Lehigh University Lehigh Preserve

Theses and Dissertations

2011

# Further Development and Application of Data Model for Large-Scale Structural Experiments

STANLEY C.H. KING Lehigh University

Follow this and additional works at: http://preserve.lehigh.edu/etd

#### **Recommended** Citation

KING, STANLEY C.H., "Further Development and Application of Data Model for Large-Scale Structural Experiments" (2011). *Theses and Dissertations*. Paper 1206.

This Thesis is brought to you for free and open access by Lehigh Preserve. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Lehigh Preserve. For more information, please contact preserve@lehigh.edu.

## FURTHER DEVELOPMENT AND APPLICATION OF DATA MODEL FOR LARGE-SCALE STRUCTURAL EXPERIMENTS

by

**Stanley King** 

A Thesis Presented to the Graduate and Research Committee of Lehigh University in Candidacy for the Degree of Master of Science

> in Civil Engineering

Lehigh University

April 2011

This thesis is accepted an approved in partial fulfillment of the requirements for the Master of Science.

April 27 2011 Date

Danse

Dr. Richard Sause Thesis Advisor

Dr. Stephen Pessiki Chairperson of Department

### Acknowledgements

I would like to express my gratitude to my research advisor, Dr. Richard Sause, for giving me this opportunity to work on this research and being patient to help me to accomplish it. With his guidance and assistance, I have increased my ability in the structural engineering field. I also would like to thank Dr. James M. Ricles, who provided me with his valued insight on experiments conducted at the Real-Time Multi-Directional (RTMD) earthquake simulation facility, and Dr. Chang-Ho Lee, who provided me with his knowledge on data modeling during the research.

I must extend many thanks to everyone at the ATLSS Center including Peter Bryan and Tommy Marullo for their assistance on my research, especially Tommy Marullo, who developed the Lehigh RTMD Metadata Web, software.

I sincerely appreciate the financial support from Lehigh University and the George E. Brown, Jr. Network for Earthquake Engineering Simulation (NEES) sponsored by the National Science Foundation (NSF).

Finally, I would like to thank my family, especially my parents and my wife, Chialin. Without their love, patience, and support, I would not have been able to achieve this. I also like to thank my friends who have provided me with their support and encouragement during this process.

## **Table of Contents**

Acknowle	dgements	iii	
Table of C	Table of Contents		
List of Fig	List of Figures		
Abstract		1	
Chapter 1	Introduction		
1	1 Overview	2	
1	2 Research Objectives	3	
1	3 Outline	4	
Chapter 2	Overview of Lehigh Model		
2	1 Introduction	5	
2	2 Notation for Classes and Attributes of Lehigh Model	5	
2	3 Development of Lehigh Model	6	
2	4 Primary Class Hierarchy of Lehigh Model	7	
2	5 Summary	11	
Chapter 3	Update of Lehigh Model		
3	1 Introduction	18	
3	2 Earlier Lehigh Model	18	
3	3 Updated Classes and Attributes of Lehigh Model	20	
	3.3.1 Project and Experimental Task Classes	20	
	3.3.2 Abstract Test Condition Class	20	

		3.3.3 Test Protocol Class	
		3.3.4 Data Set Class	
	3.3.5 Simulation Coordinator and Simulation Classes		22
	3.3.6 Analytical Substructure Computation and Physical		23
Substructure Test Classes		Substructure Test Classes	
		3.3.7 Analysis Task, Analysis, and Analysis Case Classes	24
	3.4	Summary	25
Chapte	r 4	Lehigh RTMD Metadata Web	
	4.1	Introduction	39
	4.2	Lehigh Model and Lehigh RTMD Metadata Web	39
	4.3	Development Tool of Lehigh RTMD Metadata Web	
	4.4	Brief Guide to Lehigh RTMD Metadata Web	
	4.5 Summary		43
Chapter 5		Example of Lehigh RTMD Metadata Web Applied to Typical	
		Experiment	
	5.1	Introduction	51
	5.2	Background of Project	51
	5.3	Information and Data in Lehigh RTMD Metadata Web	52
		5.3.1 Project Class and Experimental Task Class Levels	52
		5.3.2 Test Condition Class Level	53
		5.3.3 Test Class Level	57
		5.3.4 Data Set Class Level	57

5.4	Summary	58	
Chapter 6	Chapter 6 Example of Lehigh RTMD Metadata Web Applied to Hybrid		
	Simulation		
6.1	Introduction	78	
6.2	Background of Project	78	
6.3	Information and Data in Lehigh RTMD Metadata Web	79	
	6.3.1 Project Class and Experimental Task Class Levels	80	
	6.3.2 Test Condition Class Level	80	
	6.3.3 Test Class Level	82	
	6.3.4 Data Set Class Level	83	
6.4	Summary	84	
Chapter 7	Summary, Conclusions, and Future Work		
7.1	Summary	101	
7.2	Conclusions	102	
7.3	Recommendations for Future Work	102	
References		103	
Vita		105	

# List of Figures

<b>Chapter 2</b> Figure 2.1	Levels of Classes in Class Hierarchy (Adopted from Lee et al., 2006)		
Figure 2.2	Reference NEESgrid Model (Peng and Law, 2004)		
Figure 2.3	Oregon State Model (Oregon State University, 2003)		
Figure 2.4	NEEScentral Model (NEESit, 2006)		
Figure 2.5	Databases for Structural Test Data		
Chapter 3 Figure 3.1	Project, Experimental Task, and Test Condition Classes (Adopted from Lee et al., 2006)	27	
Figure 3.2	Specimen Classes (Adopted from Lee et al., 2006)	28	
Figure 3.3	Facility Class (Adopted from Lee et al, 2006)	29	
Figure 3.4	Loading Fixture and Bracing and Reaction Fixture Classes (Adopted from Lee et al., 2006)	29	
Figure 3.5	Location Class (Adopted from Lee et al., 2006)	30	
Figure 3.6	Classes for Data Acquisition and Test Control (Adopted from Lee et al., 2006)	31	
Figure 3.7	Test and Analysis Classes (Adopted from Lee et al., 2006)	32	
Figure 3.8	Classes of Hybrid Experimental Task (Adopted from Lee et al., 2006)	33	
Figure 3.9	Updated Project, Experimental Task, Analysis Task, and Test Condition Classes	34	
Figure 3.10	Updated Test, Test Protocol, and Data Set Classes	35	

Figure 3.11	Updated Classes of Hybrid Experimental Task		
Figure 3.12	Updated Analysis Class and Related Classes		
<b>Chapter 4</b> Figure 4.1	Test Condition 0 Web Page of Lehigh RTMD Metadata Web	44	
Figure 4.2	Interaction between PHP Data Model Configuration and MySQL		
Figure 4.3	Login Web Page of Lehigh RTMD Metadata Web		
Figure 4.4	Administration Web Page of Lehigh RTMD Metadata Web		
Figure 4.5	Projects Web Page of Lehigh RTMD Metadata Web	46	
Figure 4.6	Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems Web Page of Lehigh RTMD Metadata Web		
Figure 4.7	Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF System Web Page of Lehigh RTMD Metadata Web		
Figure 4.8	Test 0A Web Page of Lehigh RTMD Metadata Web	47	
Figure 4.9	Test 0A Data Set Web Page of Lehigh RTMD Metadata Web	48	
Figure 4.10	Example of Editing Description	48	
Figure 4.11	Example of Uploading Files	49	
Figure 4.12	Example of Creating a Shared Object	49	
Figure 4.13	Example of Creating a Non-shared Object	50	
Chapter 5 Figure 5.1	Experiments on PT Connection with BFFD (Adopted from Wolski, 2006)	59	
Figure 5.2	Self-Centering Damage Free Seismic-Resistant Steel Frame Systems Project Web Page	60	

Figure 5.3	Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems Web Page		
Figure 5.4	Files for Publications		
Figure 5.5	Test Condition 5 Web Page		
Figure 5.6	Specimen # 2 Web Page		
Figure 5.7	Tab E-1 of Setup Drawing File		
Figure 5.8	BFFD (Fillet Weld) Web Page	65	
Figure 5.9	Location of BFFD Web Page	65	
Figure 5.10	Geometry of BFFD Web Page	66	
Figure 5.11	Tab S-1 of Setup Drawing File		
Figure 5.12	Material of Slot Plate Web Page		
Figure 5.13	Yield Stress Web Page	67	
Figure 5.14	Material of Friction Plate Web Page	68	
Figure 5.15	Real-Time Multi-Directional (RTMD) Facility Web Page	68	
Figure 5.16	Files for Facility Specifications	69	
Figure 5.17	NEES Actuator 2 Web Page	70	
Figure 5.18	Descriptive, Specification, and Drawing Files for NEES Actuator 2		
Figure 5.19	Bracing Fixture for Specimen (Adjusted) Web Page		
Figure 5.20	LVDT-3 Web Page	73	

Figure 5.21	File of Calibration and Specification of LVDT-3		
Figure 5.22	Location of LVDT-3 Web Page		
Figure 5.23	Tab IN-2 of Setup Drawing File		
Figure 5.24	Cables Web Page		
Figure 5.25	Test 5B Web Page		
Figure 5.26	Test Protocol # 14 Web Page		
Figure 5.27	Test 5B Data Set Web Page	77	
Chapter 6 Figure 6.1	MDOF Real-time Hybrid PSD Test (Adopted from Mercan, 2007)	85	
Figure 6.2	Physical Substructure Setup (Adopted from Mercan, 2007)		
Figure 6.3	Simulink Model for Real-Time Hybrid PSD Simulation with MDOF Analytical Substructure and an Elastomeric Damper as Physical Substructure (Adopted from Mercan, 2007)		
Figure 6.4	Studies on Large-Scale, Real-Time Pseudodynamic Testing Web Page	87	
Figure 6.5	Hybrid PSD Tests with MDOF Analytical Substructure Task Web Page	87	
Figure 6.6	Files for Publications	88	
Figure 6.7	Target PC Web Page	89	
Figure 6.8	Integration Algorithm Web Page		
Figure 6.9	Subsystem 1 of Simulink Model for Real-Time Hybrid PSD Simulation – Simulation Coordinator in Highlighted Box		

Figure 6.10	3 Story Substructure Web Page		
Figure 6.11	3 DOF Frame Web Page		
Figure 6.12	Subsystem 1 of Simulink Model for Real-Time Hybrid PSD Simulation – 3 DOF Frame Analytical Substructure in Highlighted Box		
Figure 6.13	Damper Testbed Web Page	93	
Figure 6.14	Damper Specimen Web Page	94	
Figure 6.15	North A-Frame Web Page	95	
Figure 6.16	Clevis Plates Web Page	95	
Figure 6.17	Linear MDOF (KFF=2) Simulation Web Page	96	
Figure 6.18	Linear MDOF (KFF=2) Computation Web Page	97	
Figure 6.19	Linear MDOF (KFF=2) Test Web Page	98	
Figure 6.20	Linear MDOF 3 Story Substructure Web Page	98	
Figure 6.21	Linear MDOF Damper Tested Web Page	99	
Figure 6.22	Data Set Web Page for Simulation	99	
Figure 6.23	Data Set Web Page for Computation		
Figure 6.24	Data Set Web Page for Physical Test	100	

#### Abstract

The information and data related to the large-scale structural experiments are often complicated and contained in various documents, drawings, photos, and other computerbased files. A data model is needed to efficiently access, share, and use the information and data. The Lehigh Model was developed by Lee et al. (2006) at the Real-Time Multi-Directional (RTMD) earthquake simulation facility at the ATLSS Center at Lehigh University. The development of the Lehigh Model was based on a review of previous work on data models and databases for structural experiments and a study of a number of large-scale structural experiments conducted at the ATLSS Center. The primary class hierarchy of the Lehigh Model consists of the project class level, the experimental task class level, the test condition class level, the test class level, and the data set level. Based on the earlier Lehigh Model, updated versions of the classes and attributes of the Lehigh Model are presented to better represent the information and data for large-scale structural experiments conducted at the RTMD earthquake simulation facility.

The Lehigh RTMD Metadata Web is an implementation of the updated Lehigh Model. Two examples of applying the Lehigh RTMD Metadata Web are presented. The first example is the Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems project. The second example is the Studies on Large-scale, Real-time Pesudodynamic (PSD) Testing project, which focuses on hybrid earthquake simulation.

## **Chapter 1**

### Introduction

#### **1.1 Overview**

Structural researchers perform laboratory experiments to examine and understand the performance of structural components, connections, and assemblies and to develop ways of enhancing this performance. Often these experiments are conducted at large-scale on complex structural assemblies. The experiments may involve one or more test methods, including quasi-static, pseudo-dynamic, or hybrid pseudo-dynamic test methods, to closely simulate realistic loading conditions. The results from these experiments are published in papers and reports, shared with other researchers, and used for related research and practical applications.

The large amount and the diversity of the information related to structural experiments, including data files, drawing files, photos, videos, researcher's notes, and other descriptions of the test specimens, test facility, test methods, and test fixtures, make it difficult to efficiently access, share, and use the information. The relationships among the different types of information (e.g., among the test data, the drawings for different specimens, photos of different tests, and the test methods employed in different tests) are often unclear and perhaps misleading to others who may try to use the data after the experimental research project is complete. In addition, organized (structured) searches to locate specific elements of the information are often impossible. Even the research team who conducted the experiments may not be able to efficiently access all of the details of

this information. As a result, while the main concepts from the structural experiments are disseminated in papers and reports, many interesting and important details of the experiments may not be readily available or easily shared.

A data model for structural experiments can be used to organize and represent the related information and data. A data model helps researchers logically organize and manage the information and data from structural experiments using predefined hierarchies and categories of information. The logical organization of the information enables relationships among information to be established, enables missing information to be identified, and enables structured searches to locate specific elements of the information. Once the data model is implemented, the implemented model provides structural researchers with a convenient means to access, share, and use the information. If the implemented model can be accessed through the internet, then a wide range of researchers and practitioners can access and use the results from the structural experiments.

#### **1.2 Research Objectives**

This research is based on an earlier Lehigh Model developed by Lee et al. (2006). The objectives of this research are to further develop the classes and attributes of the Lehigh Model and to study and present an implementation of the Lehigh Model in a web-based data base (Marullo, 2007). This is to be accomplished by:

- To improve the classes and attributes of the Lehigh Model for large-scale structural experiments, including typical experiments and hybrid experiments.
- To present an implementation of the Lehigh Model, named the Lehigh RTMD Metadata Web, by introducing concepts of the implementation and illustrating its interface.
- To illustrate the implementation using examples of structural experiments conducted at the Real-Time Multi-Directional (RTMD) earthquake simulation facility in the ATLSS Center at Lehigh University.

#### 1.3 Outline

This thesis starts with the present chapter, which outlines the need for a data model for large-scale structural experiments. Chapter 2 presents the notation used to represent the classes and attributes of the Lehigh Model. The basic organization of the Lehigh Model is also presented. Chapter 3 presents the classes and attributes of the earlier Lehigh Model and the updated classes and attributes of the Lehigh Model. Chapter 4 presents a web application, named the Lehigh RTMD Metadata Web, which is an implementation of the Lehigh Model. Chapter 5 presents an example applying the Lehigh RTMD Metadata Web to a typical experiment and Chapter 6 present an example applying the Lehigh RTMD Metadata Web to a real-time hybrid experiment. Chapter 7 summarizes this thesis, provides conclusions, and makes suggestions for future work.

## **Chapter 2**

## **Overview of Lehigh Model**

#### 2.1. Introduction

This chapter presents the notation used to represent the classes and attributes of the Lehigh Model. The development of the Lehigh Model for large-scale structural experiments is summarized. The primary class hierarchy of the Lehigh Model is then presented.

#### 2.2. Notation for Classes and Attributes of Lehigh Model

The Lehigh Model consists of classes and attributes as shown in Figure 2.1. The classes and attributes are represented using a modified entity-relationship diagram developed for entity-based integrated design product and process models by Hong and Sause (1994). Each rectangular box shown in Figure 2.1 indicates a category of entities (referred to as an entity category or a class). Each attribute of a class is shown below the rectangular box with a horizontal bar. At end of each horizontal bar, a circle is attached to identify the value of the attribute. If the attribute is single-valued, the bar ends with an empty circle, and if the attribute is multi-valued, the bar ends with a solid circle. The value set of an attribute (the set of possible values for the attribute) is represented in square brackets. The attribute type is identified in parentheses below the bar. Attributes are classified into two main types: (1) "data-valued" attributes (DVA) whose values are alphanumeric or are otherwise indecomposable; and (2) "object entity-valued" attributes (OEVA) whose values refer to other classes.

include base attributes (B), internally derived attributes (DI), and externally derived attributes (DE). Some of these attribute types are used in the data model figures in this thesis.

#### **2.3.** Development of Lehigh Model

Previous work on data models and databases for structural experiments was considered during the development of the Lehigh Model. The Reference NEESgrid Model (Peng and Law, 2004), the Oregon State Model (Oregon State University, 2003), and the NEEScentral Model (NEESit, 2006) were reviewed (Lee et al., 2006). These models are shown in Figures 2.2, 2.3 and 2.4, respectively. These three data models provided insight into appropriate classes and attributes for the Lehigh Model. Two existing databases of structural experiment data, the SAC Design Information Database (http://www.sacsteel.org/, 2006) and the PEER Structural Performance Database enable searches, based on certain types of data, to be conducted to retrieve data from the databases. The "searchable fields" of these databases, shown in Figure 2.5 (a) and (b), suggest the types of information associated with structural experiments that should be included in the Lehigh Model.

A number of large-scale structural experiments conducted at the Real-Time Multi-Directional (RTMD) earthquake simulation facility were also studied. The information and data related to these experiments suggested the primary classes and attributes of the Lehigh Model. Based on the previous work mentioned above and studies of experiments conducted at the ATLSS Center, the Lehigh Model was formulated to represent the information and data associated with large-scale structural experiments. An earlier report on the Lehigh Model (Lee et al., 2006) explains in more detailed information how the previous work and studies of experiments were considered in the development of the Lehigh Model.

#### 2.4. Primary Class Hierarchy of Lehigh Model

The primary class hierarchy of the Lehigh Model consists of five main levels as shown in Figure 2.1, namely, the project class level, the experimental task class level, the test condition class level, the test class level, and the data set class level. The information and data for experiments are classified into these class levels in the Lehigh Model. The definitions of the class levels and the relationships between the class levels are described below:

#### **Project Class Level**

The project class level is at the highest level in the Lehigh Model. This level includes the project class to represent fundamental information about structural research projects, such as objectives, project scope, summarized conclusions, associated organizations, and the experiments and/or analyses performed. As shown in Figure 2.1 a project class has two multi-valued attributes for experimental tasks and analysis tasks, respectively.

#### **Experimental Task Class Level**

This level represents the experiments and the analyses conducted under a specific project. Two main classes are included at this level as shown in Figure 2.1, namely, the experimental task class and the analysis task class, which are the value sets of the attributes of the project class. Information related to experimental tasks and analytical tasks are included in these classes.

The experimental task class is a generalization of the typical experimental task class and the hybrid experimental task class. The typical experimental task class represents experiments using typical methods for structural tests, such as the quasi-static method and the pseudo-dynamic method. The typical experimental task class shown in Figure 2.1 has a multi-valued attribute for test conditions to represent the experimental setup. The hybrid experimental task class represents experiments using the hybrid pseudo-dynamic method; either a local hybrid experiment or a distributed hybrid experiment. The hybrid experimental task class shown in Figure 2.1 includes a single-valued attribute for simulation coordinator and two multi-valued attributes for analytical substructures and physical substructures, which represent hybrid experimental setups based on the substructuring technique (Dermitzakis and Mahin, 1985).

The analysis task class represents numerical simulations which are independent of any experimental task under a project. The analysis task class shown in Figure 2.1 includes a multi-valued attribute for analysis conditions to represent the details of analyses that are conducted.

#### **Test Condition Class Level**

This level represents the details of experimental setups, which includes physical and analytical setups. Because of the varied test methods considered, an experiment task (as well as an analysis task) consists of one or more different types of experimental setups.

The test condition class is the value set of the test conditions attribute of the typical experimental task class as shown in Figure 2.1. The test condition class represents the laboratory conditions for typical experiments. The class includes information on the specimen, facility, loading fixtures, bracing and reaction fixtures, sensors, and cables which are the main components of experimental setups. A multi-valued attribute for tests is included in the test condition class to represent the test protocols and resulting test data.

Because of the complexity of a hybrid experimental task, three classes are need to represent the experimental conditions, namely, the simulation coordinator class, the analytical substructure class, and the physical substructure class, which are the value sets of the attributes of the hybrid experimental task class as shown in Figure 2.1. The simulation coordinator class represents information on the simulation model, software, and simulation facility used to run the simulation. The analytical part of a hybrid experimental task is represented by the analytical substructure class which includes information on the analytical substructure model, software, and facility used to run the analytical simulation. The physical part of a hybrid experimental task is represented by the physical substructure class which includes information on the laboratory conditions which is similar to the test condition class. Three multi-valued attributes for simulations, analytical substructure computations, and physical substructure tests are included in the simulation coordinator class, the analytical substructure class, and the physical substructure class, respectively, to represent the data from a hybrid simulation.

In Figure 2.1, the analytical conditions of an analysis task are represented by the analysis class, which is the value set of the analysis conditions attributes of the analysis task class. The analysis class includes information on the software and model used in the analysis. The analysis class has a multi-valued attributes for analysis cases to represent the data from the analyses.

#### **Test Class Level**

This level represents loading conditions, such as history files, simulation method files, data acquisition system configuration files, controller configuration files, and input files for experiments and analytical simulations. In Figure 2.1, the classes at this level are the test class, the simulation class, the analytical substructure computation class, the physical substructure test class, and the analysis case class. These classes are the value sets of the attributes of the test condition class, the simulation coordinator class, the analytical substructure class, the physical substructure class, and the analysis class, and the analysis class, respectively. All of these classes have a single-valued attribute to represent either the experimental data set or the analytical data set resulted from the experiment or analysis.

#### **Data Set Level**

The results of experiments and analytical simulations are represented at this level. The abstract data set class in Figure 2.1 represents common attributes of the experimental data set class and the analytical data set class. As shown in Figure 2.1, the experimental data set class includes raw data and corrected data from a test or from a physical substructure test; the analytical data set includes output data from a simulation, an analytical substructure computation, or an analysis case.

#### 2.5. Summary

This chapter presents an overview of the Lehigh Model. The notation used for the classes and attributes of the Lehigh Model is described. The development of the Lehigh Model is based on previous work on data models and databases for structural experiments, as well as studies of experimental projects conducted at the RTMD earthquake simulation facility. The class hierarchy of the Lehigh Model includes five main levels, namely, the project class level, the experimental task class level, the test condition class level, the test class level, and the data set class level. These class levels are described. This chapter only presents on part of the classes and attributes of the Lehigh Model are presented in Chapter 3 and in the report by Lee et al. (2006).





Figure 2.1 Levels of Classes in Class Hierarchy (Adopted from Lee et al., 2006) (Continued)



Figure 2.2 Reference NEESgrid Model (Peng and Law, 2004)



Figure 2.3 Oregon State Model (Oregon State University, 2003)



Figure 2.4 NEEScentral Model (NEESit, 2006)



#### **Search Connections Database**

Beam Size ANY V	FEMA 350 Type	Connection Type
Column Size ANY exactly Search for in All fields Investigator ALL Return a maximum of 25 results	ALL BB BFP BSEP DST FF RBS SP SW WFP WUF-B WUF-W	All Pre-Northridge PreNR-NT PreNR-EndPl PreNR-EndPl PreNR-eventay Cover Plate End Plate End Plate Haunch Pipe Reduced Beam Section Rib Tree Side Plate

(a) SAC Design Information Database (http://www.sacsteel.org/, 2006)

Author Column Type Test Configuration Span-to-Depth Ratio Axial Load Ratio Longitudinal Reinf Ratio	- (range 0-10) <u>histogram</u> - (range -0.1-0.9) <u>histogram</u> - (range 0.002-0.06) <u>histogram</u>	Failure Type Damage Concrete Crushing Observation Significant Spalling Long Bar Buckling Long Bar Fracture Sparia Fracture Loss of Axial Load Capacity Search Clear		
For the ratios, enter a range of values to search in combination with other column attributes or view the histogram showing the distribution of values in the database and click any bar to view record details.				
Download				
Download tab-delimited data files for rectangular or spiral columns (Save as text.) Column headers are on the first row. Notation is documented on the about page.				
Rectangular columns : <u>Properti</u> Spiral columns : <u>Properties</u>   D	<u>es</u>   Damage amage			
The Structural Performance Database is a project of the <u>Parific Earthquake Engineering Research Center</u> , funded by the <u>National Science Foundation</u> . Contact <u>Marc Eberhard</u> at the University of Washington with questions or comments. Report technical problems to the <u>server administrator</u> . Copyright 2003.				

(b) PEER Structural Performance Database (http://nisee.berkeley.edu/spd/, 2006)

Figure 2.5 Databases for Structural Test Data

## Chapter 3

## **Update of Lehigh Model**

#### **3.1.** Introduction

This chapter presents a brief introduction of the earlier Lehigh Model, which was developed by Lee et al. (2006). The recently updated classes and attributes of the Lehigh Model are then presented.

#### 3.2. Earlier Lehigh Model

An earlier version of the Lehigh Model was developed by Lee et al. (2006). This earlier Lehigh Model mainly focused on typical and hybrid experiments. The classes and attributes of this earlier model are displayed in Figure 3.1 to 3.8. Figure 3.1 shows the project class, which is at the highest level of the model to represent fundamental information on a structural research project. The value set of the experimental tasks attribute of the project class refers to the experimental task class, which is a generalization of the typical experimental task class and hybrid experimental task class as shown in Figure 3.1. These two classes represent information on the typical and hybrid experiments.

The test condition class shown in Figure 3.1 is the value set of the test conditions attribute of the typical experimental task class. Because the test condition class for typical experiments and the physical substructure class for hybrid experiments have common attributes to represent the major components of laboratory conditions, an

abstract class, named the abstract test condition class shown in Figure 3.1, is created to include common attributes inherited from the test condition class and the physical substructure class. Related classes as shown in Figure 3.2 to 3.6, such as the specimen class, the facility class, the loading fixture class, the bracing and reaction fixture class, the sensor class, and the cable class, are the value sets of the attributes of the abstract test condition class. In Figure 3.1, the value set of the tests attribute of the test condition class refers to the test class shown in Figure 3.7. The test class has attributes to represent the test protocol and the resulting data.

Figure 3.8 shows the primary classes to represent information for a hybrid simulation. The simulation coordinator class, the analytical substructure class, and the physical substructure class represent part of a hybrid experimental setup. For each hybrid experimental setup, a number of simulations, analytical substructure computations, and physical substructure tests can be made as shown in Figure 3.8. The relationship among a simulation, related analytical substructure computations, and related physical substructure tests for a hybrid simulation is represented by the simulation substructure interface class, which is the value set of the corresponding attributes of the simulation class, the analytical substructure computation class, and the physical substructure test class as shown in Figure 3.8.

#### **3.3.** Updated Classes and Attributes of Lehigh Model

#### **3.3.1.** Project and Experimental Task Classes

Updated project and experimental task classes are presented in Figure 3.9. The organizations attribute is moved from the experimental task class shown in Figure 3.1 into the updated project class shown in Figure 3.9 because a project is sponsored and conducted by a number of organizations. An attribute for analyses is moved from the abstract test condition shown in Figure 3.1 into the updated experimental task class shown in Figure 3.9 as a common attribute for typical experimental tasks and the hybrid experimental tasks so that the analyses attribute can describe one or more numerical analyses that correspond to a structural experiment with a specific experimental setup and loading method. In this new location, numerical analyses used to design the experiment can also be represented by this analyses attribute.

#### **3.3.2.** Abstract Test Condition Class

Figure 3.9 also displays the updated attributes of the abstract test condition class. The attributes for DAQ systems, controller systems, and simulation systems in the abstract test condition class shown in Figure 3.1 are removed, since the configuration files for these systems may be different for various tests performed in a given experimental setup. These attributes are moved to the test protocol class described in the following section. The attribute for analyses in the abstract test condition class shown in Figure 3.1 is moved to the updated experimental task class shown in Figure 3.9 because changes in laboratory conditions may not affect the numerical simulation (i.e., a numerical simulation can represents a number of different test conditions), and also some analyses

used to design an experiment may not correspond to a specific laboratory condition. The rest of the attributes of the test condition class are the same as described in the earlier report (Lee et al., 2006).

#### **3.3.3.** Test Protocol Class

The updated test protocol class is presented in Figure 3.10 to represent the loading conditions for an experiment. The test protocol class is the value set of the test protocol attribute of the test class shown in Figure 3.10 for typical experiments, and is also the value set of corresponding attributes of the simulation class, the analytical substructure computation class, and the physical substructure test class as shown in Figure 3.11 for hybrid experiments. As shown in Figure 3.10, the attributes for the DAQ configuration files and controller configuration files are included in the updated test protocol class to replace the attributes for DAQ systems and controller systems in the abstract test condition class shown in Figure 3.1, because these configuration files, such as the DAQ calibration parameters and controller parameters, may vary for each test protocol. The other configuration files attribute is added in the updated test protocol class to represent other configuration data, such as telepresence and scramnet memory configurations of the Real-Time Multi-Directional (RTMD) earthquake simulation facility.

#### **3.3.4.** Data Set Class

Figure 3.10 shows the updated data set class, which is modified from the data set class shown in Figure 3.7 to record experimental results from physical experiments and analytical simulations. The updated data set class shown in Figure 3.10 is an abstract

class of the analytical data set class and the experimental data set class. The analytical data set class is the value set of corresponding attributes of the simulation class, the analytical substructure computation class, and the analysis case class as shown in Figure 3.11. The experimental data set class is the value set of corresponding attributes of the test class and the physical substructure test class as shown in Figure 3.10 and 3.11. The data set class shown in Figure 3.10 includes common attributes inherited from the analytical data set class and the experimental data set class. These common attributes are description, description files, processed data, images, and videos. The analytical data set class includes an attribute for output data to represent the results from analytical simulations. The experimental data set class includes attributes for raw data, corrected data, and web cams to represent the results from physical experiments. The value set of the raw data attribute refers to the raw data class, which includes attributes for binary data and engineering unit data as shown in Figure 3.10. The attribute for corrected data represents the experiment data that must be corrected or revised to compensate for calibration problems, to eliminate noise, to handle zero-offset conditions, or to apply an overall correction factor. The value set of the web cams attribute refers to the web cams class, which includes the attributes for captured images, timelapsed videos, and composite videos as shown in Figure 3.10.

#### **3.3.5.** Simulation Coordinator and Simulation Classes

Figure 3.11 shows the updated simulation coordinator class. The simulation coordinator represents the software and computation facility used to conduct/control a hybrid experiment. In order to generate commands to and receive feedback from substructures,

a numerical model is included in a simulation coordinator to run the simulations. The attributes for simulation model and software are added into the updated simulation coordinator class as shown in Figure 3.11. These two attributes describe the numerical model and software used to run the simulations. The value sets of the simulation model and software attributes refer to the model class and the software class shown in Figure 3.13, respectively. These classes are described in the updated analysis class below.

Figure 3.11 also shows the updated simulation class, which is the value set of the simulation attribute of the simulation coordinator class. The descriptive files and persons attributes are added in this updated class as shown in Figure 3.11. The attribute for descriptive files enable researches to describe a simulation by using images or existing documents; the attribute for persons is to record the people who conduct the simulations. As shown in Figure 3.11, the value set of the simulation protocol attribute is changed to refer to the test protocol class, because the test protocol class in Figure 3.10 can also represents the loading conditions used to run the simulations. The value set of the simulation data set attribute is also changed to refer to the analytical data set class in Figure 3.10.

## 3.3.6. Analytical Substructure Computation and Physical Substructure Test Classes

Figure 3.11 also shows the updated analytical substructure computation class and the updated physical substructure test, which are the value sets of the attributes of the analytical substructure class and the physical substructure class, respectively. The

attributes for descriptive files and persons are also added in these updated classes. As shown in Figure 3.11, the value set of the computation protocol attribute of the analytical substructure computation class is changed to refer to the test protocol class in Figure 3.10 to represent the loading conditions used to run the computations. The value sets of attributes for the data set of the analytical substructure computation class and the physical substructure test class are changed to refer to the analytical data set class and the experimental data set class, respectively.

#### 3.3.7. Analysis Task, Analysis and Analysis Case Classes

The analysis task class is the value set of the analysis tasks attribute of the project class as shown in Figure 3.9. The analysis task class represents numerical simulations for a project, which are independent of other experimental tasks under a project. The class includes attributes for description, descriptive files, publications, and presentations to represent information on an analysis task. A multi-valued attribute for analysis conditions is included in the analysis task class to represent various types of numerical simulations and results.

The updated analysis class shown in Figure 3.12 is modified from the analysis class in Figure 3.7 to describe details of an analytical setup, which include numerical models, computer hardware, and software. The analysis class is the value set of the analysis conditions attribute of the analysis task class and is also the value set of the analyses attributes of the experimental task class as shown in Figure 3.9. As shown in Figure 3.12, the model class represents a numerical model by description, model files, and model
images; the hardware class describes the computer system; the software class describes the software, such as OpenSees, DRAIN -2DX, SAP2000, ABAQUS, and others. These classes are the value sets of the attributes of the analysis class as shown in Figure 3.12. A multi-valued attribute for analysis cases is included in the analysis class, so a number of analysis cases can be executed for a given analytical setup.

Figure 3.12 also shows the analysis case class, which is the value set of the analysis cases attribute of the analysis class. The analysis case class represents a single numerical simulation. The attributes represent the analysis case protocol and the resulting analytical data set. As shown in Figure 3.12, the case protocol class is the value set of the case protocol attribute. The input files attribute of the case protocol class describes the loading conditions for the numerical simulation. The value set of the data set attribute of the analysis case class refers to the analytical data set class as shown in Figure 3.10 to represent the results from a numerical simulation.

## 3.4. Summary

This chapter presents the earlier Lehigh Model developed by Lee et al. (2006). The earlier model mainly focuses on typical and hybrid experiments. A concise introduction is presented to describe the major classes and attributes of the earlier model. The recently updated classes and attributes of the Lehigh Model are presented. The updated classes are the project class, the experimental task class, the abstract test condition class, the test protocol class, the data set class, the analytical data set class, the experimental data set class, the simulation coordinator class, the simulation class, the analytical substructure computation class, the physical substructure test class, the analysis task class, the analysis class, the analysis case class, and the case protocol class. Details of the explanation for unchanged classes and attributes of the Lehigh Model can be found in Chapter 4 of the earlier report (Lee et al., 2006).



Figure 3.1 Project, Experimental Task, and Test Condition Classes (Adopted from Lee et al., 2006)



Figure 3.2 Specimen Classes (Adopted from Lee et al., 2006)



Figure 3.3 Facility Class (Adopted from Lee et al, 2006)





Figure 3.4 Loading Fixture and Bracing and Reaction Fixture Classes (Adopted from Lee et al., 2006)



Figure 3.5 Location Class (Adopted from Lee et al., 2006)



sensor

cable

(Adopted from Lee et al., 2006)



Figure 3.7 Test and Analysis Classes (Adopted from Lee et al., 2006)



Figure 3.8 Classes of Hybrid Experimental Task (Adopted from Lee et al., 2006)



Figure 3.9 Updated Project, Experimental Task, Analysis Task, and Test Condition Classes



Figure 3.10 Updated Test, Test Protocol, and Data Set Classes



Figure 3.11 Updated Classes of Hybrid Experimental Task



Figure 3.11 Updated Classes of Hybrid Experimental Task (Continued)



Figure 3.12 Updated Analysis Class and Related Classes

# Chapter 4

# Lehigh RTMD Metadata Web

## 4.1. Introduction

This chapter presents the Lehigh RTMD Metadata Web, which is a web application for an implementation of the Lehigh Model. This application was programmed by Marullo (2007) based on the Lehigh Model described by Lee et al. (2006) and on the model described in this thesis.

### 4.2. Lehigh Model and Lehigh RTMD Metadata Web

A web application was developed at the Real-Time Multi-Directional (RTMD) earthquake simulation facility in the ATLSS Center, and is named the Lehigh RTMD Metadata Web (Marullo, 2007). The Lehigh RTMD Metadata Web is an implementation of the Lehigh Model, which allows users to store, access, retrieve, and search information and data associated with large-scale structural experiments. The relationship between the Lehigh Model and the Lehigh RTMD Metadata Web is described below.

The objects of the classes of the Lehigh Model are represented by the web pages of the Lehigh RTMD Metadata Web. For example, the Test Condition 0 web page shown in Figure 4.1 represents an object of the test condition class shown in Figure 3.9. The attributes of an object and their values are represented in each web page of the Lehigh RTMD Metadata Web. If the value set of an attribute refers to alphanumeric data, such as "STRING", this value is directly displayed at top of a web page, for example, the

description attribute of the Test Condition 0 web page shown in Figure 4.1. If the value set of an attribute refers to "FILE", this value is displayed in a rectangular box as a link to access the file, for example, the setup drawings and photos attribute of the Test Condition 0 web page shown in Figure 4.1. If the value set of an attribute refers to another object, this value is presented in a rectangular box as an object link to another web page that corresponds to the class to which the value set refers in the Lehigh Model, for example, the specimen attribute of the Test Condition 0 web page shown in Figure 4.1. A multi-valued attribute is indicated by a plus sign at the end of the rectangular box, for example, the setup drawings and photos attribute as shown in Figure 4.1. A single-valued attribute does not have a plus sign at end after the value is input, for example, the specimen attribute as shown in Figure 4.1.

## 4.3. Development Tool of Lehigh RTMD Metadata Web

The Lehigh RTMD Metadata Web was developed by Marullo (2007) using PHP and Javascript to produce HTML for web browsers such as Firefox. It interfaces with MySQL, a database system, to store and retrieve data. The Lehigh RTMD Metadata Web reads a user-generated PHP data model schema for a project. This schema provides the hierarchy of the objects from the five main levels of the Lehigh Model to be represented by the Lehigh RTMD Metadata Web, which then presents an interface, as well as web pages as shown in Figure 4.1, for a user to create the objects, to view the objects and to edit or delete the objects as described in the following section 4.4. A PHP data model configuration defines created object names and lists direct child objects and shared objects for interacting with the MySQL database. The MySQL database

contains a generic template table which allows any type of object to be instantiated. This new row of data contains the object data along with a reference to its unique parent ID. Another template table exists to maintain one-to-many (1-M) and many-to-many (N-M) object relationships, called shared objects in the Lehigh RTMD Metadata Web. Figure 4.2 shows the interaction between the PHP data model configuration and MySQL database to enable the web application to view objects and related data. This implementation allows the data model to be easily changed without having to modify the database structure.

## 4.4. Brief Guide to Lehigh RTMD Metadata Web

The Lehigh RTMD Metadata Web has security to protect experimental resources, as shown in Figure 4.3; it requires users to login. The administration web page shown in Figure 4.4 allows users to request membership to a specific project to view or update the data. All of the projects in the Lehigh RTMD Metadata Web database are listed in the Projects web page shown in Figure 4.5. If the project name is displayed in black, the user has authority to access that project; if the project name is displayed in gray, the user is not permitted to access that project. Users are permitted to create their own projects by clicking on the "Add Project" button under the listed projects.

A project, called the Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems project, is selected as an example to present the Lehigh RTMD Metadata Web. The objects related to the five main levels of the Lehigh Model for the project are shown in Figures 4.6, 4.7, 4.1, 4.8, and 4.9. Figure 4.6 shows the Self-Centering Damage-Free

Seismic-Resistant Steel Frame System project web page as an object of the project class at the project class level. Figure 4.7 shows the Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems web page as an object of the typical experimental task class at the experimental task class level. Figure 4.1 shows the Test Condition 0 web page as an object of the test condition class at the test condition class level. Figure 4.8 shows the Test 0A web page as an object of the test class at the test class level. Finally, Figure 4.9 shows the Test 0A data set web page as an object of the data set class at the data set class level. A lower level object is accessed by clicking the object link in the rectangular box that corresponds to the attribute representing the lower level object; a higher level object can be reached by clicking on the back arrow button at bottom of the web page.

The information and data for each object is shown on each web page in boxes that correspond to the attributes of the object. When the value set of an attribute refers to "STRING" or "NUMBER", this value can be input or modified in the text area as shown in Figure 4.10 by clicking on the "Edit" button on the bottom of a web page as shown in Figure 4.1. When the value set of an attributes refers to "FILE", a file can be uploaded via the web page as shown in Figure 4.11 by clicking on the plus sign button at the right end of an attribute box on a web page as shown in Figure 4.1. When the value set of an attribute box on a web page as shown in Figure 4.1. Figure 4.13 illustrate how to create an object link to a shared object and a non-shared object, respectively. A shared object is introduced in the Lehigh RTMD

Metadata Web application to allow an object to be shared between other projects, because the same object may be used in different projects, for instance, the actuator in the Test Condition 0 web page shown in Figure 4.1. A specific actuator at the RTMD earthquake simulation facility can be used for many experiments in the facility. As shown in Figure 4.12, an object link for the actuator can be created by selecting an existing, shared object of NEES Actuators or by adding a new actuator object which permits this new object to be shared with other projects. Figure 4.13 shows the web page for creating a non-shared object which is a unique object in a project, for instance, the tests in the test condition 0 web page shown in Figure 4.1, because each test represents a unique experiment. In order to present each web page with clarity, all file and object links are hidden inside the rectangular boxes. When the text in the rectangular box is displayed in bold type, it means that there are files or object links hidden in the rectangular box. These files or object links are displayed by clicking on the triangle sign at the left side of the rectangular box, as shown in Figure 4.1.

### 4.5. Summary

The Lehigh RTMD Metadata Web (Marullo, 2007) was developed to implement the Lehigh Model, and to enable the information and data associated with large-scale structural experiments to be stored, accessed, and searched. The web pages of the Lehigh RTMD Metadata Web represent objects of the classes of the Lehigh Model; the rectangular boxes on the web pages represent the attributes of an object. PHP and Javascript code, interfacing with MySQL to store and retrieve data, were used to develop this web application.

			EHIGH UNIVERSITY.
SELF-CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEM	s L'L	chc305@lehigh.edu [logout]
Projects	Test Condition: Test Condition U		
Search (beta)	Name: Test Condition 0 Description: Test condition 0 was us seat the beam, and wo The area of interest of t	ed to rk out a he con	evaluate the operation of the instrumentation, any issues with the test setup. nection is near the beam-colum [more]
	► Descriptive Files	٢	► Setup Drawings and Photos ③
	► Persons	٢	▼ Specimen Specimen #1 ×
		1	
	► Facility		► Actuators
	► Other Loading Fixtures	0	▶ Bracing and Reaction Fixtu ③
	► Sensors	٢	► Cables ③
	► Tests	٢	
	🛃 Edit		
	× D elete		
	- Back To Typical Experimental Tas	k	
		LEHIGH	UNIVERSITY RTMD METADATA WEB (VERSION 1.2)

Figure 4.1 Test Condition 0 Web Page of Lehigh RTMD Metadata Web



Figure 4.2 Interaction between PHP Data Model Configuration and MySQL



Figure 4.3 Login Web Page of Lehigh RTMD Metadata Web

WELCOME	chc305@lehigh.edu [logout]
Projects	Welcome
	Welcome to Version 1.2 of the Lehigh University RTMD Metadata Web. Lehigh users can use their existing Lehigh email to log in without registering. If you do not have a Lehigh email, you need to register by clicking the Not Registered link on the left. Once logged in, click on "Projects" in the left panel to view projects you are a member of.
	Please see the section below on requesting view or admin membership to existing projects.
	It is recommended that you use Firefox to view this page.
	Enter Project name(s) to request view or admin membership to:
	LENTCH UNIVERSITY RTMD METADATA WER (VERSION 1-2)

Figure 4.4 Administration Web Page of Lehigh RTMD Metadata Web



Figure 4.5 Projects Web Page of Lehigh RTMD Metadata Web

RTMD		UNIVERSITY.	
SELF-CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]	
Projects	Project: Self-Centering Damage-Fr Steel Frame Systems	ee Seismic-Resistant	
Search (beta)	Name:         Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems           Description:         New earthquake-resistant structural steel moment resisting frame (MRF) systems are being developed by a research group led by Lehigh University in collaboration with Princeton and Purdue Universities [more]		
	Descriptive Files     O     Publicat	ions 💿	
	► Presentations		
	▼ Typical Experimental Tasks ② ► Hybrid	Experimental Tasks	
	Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems ↓	s Tasks 📀	
	☑ Edit X D elete		
	LEHIGH UNIVERSITY	( RTMD METADATA WEB (VERSION 1.2)	

Figure 4.6 Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems Web Page of Lehigh RTMD Metadata Web

		EEHIGH	
SELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]	
Projects	Typical Experimental Task: Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems		
🔍 Search (beta)	Name:         Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems           Description:         This experimental task evaluates the development of an energy dissipation (ED) device for earthquake resistant self-centering steel moment resistin frames (SC-MRFs) with post-tensioned steel moment c [more]		
	Descriptive Files     Presentations	ublications O	
	► Test Condition	nalyses 🧿	
	EHIGH UNIX	VERSITY RTMD METADATA WEB (VERSION 1.2)	

Figure 4.7 Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF System Web Page of Lehigh RTMD Metadata Web

SELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Test: Test OA	
Search (beta)	Name: Test 0A Description: Test 0A was used to eval seat the beam, and work Start Date and Time: Dec 13 2005 9:00 am End Date and Time: Dec 13 2005 9:20 am Uncontrolled Test Condition:	luate the operation of the instrumentation, cout any issues with the test setup.
	Descriptive Files	Persons
	▼ Data Set Test 0A data set	
	Edit X Delete	
	Back To Test Condition	

Figure 4.8 Test 0A Web Page of Lehigh RTMD Metadata Web

				GH
SELF-CENTERING DAMAGE-FR	EE SEISMIC-RESISTANT STEEL FRAME SYSTEM	\$	chc305@lehigh.edu [	[logout]
Projects	Data Set: Test 0A data	set		
🔍 Search (beta)	Name: Test OA data set Description:			
	► Descriptive Files	0	► Raw Binary Data	
	▶ Raw Engineering Unit Data	0	Corrected Data	
	► Processed Data	٢	► Images	
	► Videos	٢	► Webcam Images	•
	► Timelapsed Video	٢	Composite Video	0
	► Other Files	٢		
	Edit			
	🖙 Back To Test			
		LEHIG	H UNIVERSITY RTMD METADATA WEB (VERSI	ON 1.2)

Figure 4.9 Test 0A Data Set Web Page of Lehigh RTMD Metadata Web



Figure 4.10 Example of Editing Description



Figure 4.11 Example of Uploading Files

RTMD		EEHIGH UNIVERSITY.
SELF-CENTERING DAMAGE	-FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Add Actuator	
	Name:       NEES Actuator 1         NEES Actuator 2       NEES Actuator 3         NEES Actuator 4       NEES Actuator 5         Add Link(s)       Image: Constant of the second sec	
	LEHIGH UNIVE	ERSITY RTMD METADATA WEB (VERSION 1.2)

Figure 4.12 Example of Creating a Shared Object



Figure 4.13 Example of Creating a Non-shared Object

## Chapter 5

# Example of Lehigh RTMD Metadata Web Applied to Typical Experiment

## 5.1. Introduction

An experimental task of the Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems project is used as an example of applying the Lehigh RTMD Metadata Web to a typical experiment. The information and data associated with this experiment are organized using the Lehigh RTMD Metadata Web. The information and data are presented in the five main levels of the Lehigh Model (the project class level, the experimental task class level, the test condition class level, the test class level, and the data set class level).

## 5.2. Background of Project

The purpose of the project, Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems, is to investigate steel moment resisting frames (MRFs) with posttensioned steel moment connections (PT connections) (Ricles et al., 2001) under earthquake loading. One of the experimental tasks of the project, which included a number of structural tests conducted at the RTMD earthquake simulation facility at the ATLSS Center, is to evaluate energy dissipation (ED) devices for earthquake resistant self-centering steel moment resisting frames (SC-MRFs) with PT connections (Wolski, 2006). The information and data associated with this experimental task are used to present the application of the Lehigh RTMD Metadata Web to a typical experiment. The basic experimental setup is displayed in Figure 5.1. An exterior PT steel MRF connection with a bottom flange friction device (BFFD) (Wolski, 2006) from the prototype structure shown in Figure 5.1(a) was investigated. The prototype beam section in this experimental task is an A572 Grade 50 W36x300 section. This section was scaled by 3/5 resulting in an A572 Grade 50 W21x111 section in the test specimen shown in Figure 5.1(b). The beam in the test specimen is oriented to the vertical position for convenience in the lab. The connection was designed to replicate the elongation of the strands in the prototype structure, where the elongation of each PT strand is approximately the same. In order to accomplish this, the strands are concentrated at the centroid of the beam section as shown in Figure 5.1(b). Figure 5.1(c)

### 5.3. Information and Data in Lehigh RTMD Metadata Web

The information and data associated with the selected experimental task has been organized using the Lehigh RTMD Metadata Web. Only selected web pages and related files are presented for this application example. The information and data are presented in the five main levels of the Lehigh Model, which are the project class level, the experimental task class level, the test condition class level, the test class level, and the data set class level.

### 5.3.1. Project Class Level and Experimental Task Class Levels

The fundamental information on the project can be found in the Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems Project web page shown in Figure 5.2, which represents an object of the project class of the Lehigh Model. The description describes the overall project and its research objectives. Files for related publications and presentations associated with this project are contained in the publications and presentations attributes. The object link in the typical experimental task attribute provides access to the Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems web page shown in Figure 5.3, which represents an object of the typical experimental task class of the Lehigh Model.

A description of this experimental task is shown in Figure 5.3. The Test Matrix in the descriptive files attribute provides the configuration of the tests for this experimental task. The BFFD Status Report 1, the thesis by Wolski (2006), and the 4ICEE conference paper shown in Figure 5.4 (a) through (c), respectively, are contained in the publications attribute. The presentation by Wolski (2006) is available from the presentation attribute. A number of object links in the test condition attribute provide access to each test condition used in this experimental task.

## 5.3.2. Test Condition Class Level

As defined in the Lehigh Model, a typical experimental setup consists of several major components, which are the specimen, facility, sensors, loading fixtures, bracing and reaction fixtures, and cables. Information on these components for a typical experimental setup can be found in corresponding web pages, such as the Test Condition 5 web page shown in Figure 5.5. This web page represents an object of the test condition class of the Lehigh Model. The Test Condition 5 web page is one of nine

experimental setups used in this experimental task as shown in Figure 5.3. The other test conditions are accessed from the object links in the test conditions attribute shown in Figure 5.3.

A description of this experimental setup is given in the Test Condition 5 web page shown in Figure 5.5. An AutoCAD file, named setup drawings, can be accessed from the setup drawings and photos attribute to provide an illustration of the experimental setup. The information about the related specimen, facility, loading fixtures, bracing and reaction fixtures, sensors, and cables can be found in the web pages accessed from the object links in the corresponding attributes as shown in Figure 5.5. A number of tests performed using this experimental setup can be accessed from the object links in the tests attributes shown in Figure 5.5.

Specimen # 2 is the test specimen used for this experimental setup and its detailed information is represented on the Specimen # 2 web page shown in Figure 5.6. This web page represents an object of the specimen class of the Lehigh Model. The description contains information about Specimen # 2. The AutoCAD file from the specimen drawings and photos attribute provides an illustration of the specimen as shown in Figure 5.7. The information about specimen components are found in the web pages accessed from the object links in the specimen components attribute. The BFFD (Fillet Weld) web page shown in Figure 5.8 represents one of specimen components used in Specimen # 2. The detailed information about this component is represented in the description, location, geometry, and material attributes. The Location of BFFD web

page shown in Figure 5.9 provides access to geometric location data for the BFFD, which is in Tab-E1 of the setup drawing file in the drawings and photos attribute as shown in Figure 5.7. The Geometry of BFFD web page shown in Figure 5.10 provides access to geometry information for the BFFD, which is in Tab-S1 of the setup drawing file as shown in Figure 5.11. Each specimen component has its own material information as shown in the BFFD (Fillet Weld) web page. For instance, the Material of Slot Pate web page shown in Figure 5.12 represents the steel material information for this specimen component with a steel type, elastic modulus, yield stress, ultimate stress, and material property files. Figure 5.13 shows the yield stress with the nominal value and the average material test value. Information on materials which are not standard steel, concrete, or reinforcing bars, are represented using a web page for other material, for example, the Material of Friction Plate web page shown in Figure 5.14.

The Real-Time Multi-Directional (RTMD) Facility web page shown in Figure 5.15 represents an object of the facility class of the Lehigh Model. The general information about this facility used for this experimental setup is provided in the description. The specification files in the specifications attribute provide more detailed information about the equipment used in the RTMD earthquake simulation facility as shown in Figure 5.16.

NEES Actuator 2 is part of loading fixtures used in the experimental setup. Information about this actuator can be found on the NEES Actuator 2 web page shown in Figure 5.17, which represents an object of the actuator class of the Lehigh Model. The descriptive

files, specifications, and drawings and photos attributes provide detailed information about this actuator, as shown in Figure 5.18.

The bracing and reaction fixture used in the experimental setup is represented in the Bracing fixture for specimen (adjusted) web page shown in Figure 5.19, which represents an objection of the bracing and reaction fixture class of the Lehigh Model. The description provides information about the components of this bracing and reaction fixture. An illustration of this bracing and reaction fixture is provided by the AutoCAD file in the fixture drawings and photos attributes as shown in Figure 5.7.

The LVDT-3 web page shown in Figure 5.20 represents an object of the sensor class of the Lehigh Model. Figure 5.21 shows the calibration and specification files for this sensor. These files can be accessed from the calibration and specifications attributes shown in Figure 5.20. The object link in the location using drawings and photos attributes provides access to the Location of LVDT-3 web page shown in Figure 5.22 which indicates the location information using the AutoCAD file in the drawings and photos attribute as shown in Figure 5.23.

Figure 5.24 shows the Cables web page, which represents an object of the cable class of the Lehigh Model. The information about the cables was not recorded when the experiment was performed, so all of the cables are described on one web page with the types and number of cables that were used in the experimental setup.

### 5.3.3. Test Class Level

As defined in the Lehigh Model, a test condition can include more than one test. As shown on the Test Condition 5 web page (Figure 5.5), three tests were performed using this experimental setup. The information about these tests can be found on web pages, such as the Test 5B web page shown in Figure 5.25. This web page represents an object of the test class of the Lehigh Model. This web page includes a brief description about Test 5B. The information on the loading protocol for this test is represented on the Test protocol # 14 web page shown in Figure 5.26, which can be accessed from the object link in the test protocol attribute shown in Figure 5.25. The description describes the loading method and important parameters of this loading protocol. The history file attribute provide information about the loading history used for this test. Only the simulation and DAQ configuration files are available for this test and these files are accessed by the simulation method files and DAQ config files attributes shown in Figure 5.26. The results of this test can be accessed from the object link in the data set attribute shown in Figure 5.25.

### 5.3.4. Data Set Class Level

The test results for Test 5B are represented on the Test 5B data set web page shown in Figure 5.27, which represents an object of the experimental data set class of the Lehigh Model. The description and descriptive files contain information about the resulting data. The data is accessed from the raw data engineering unit attribute, the corrected data attribute, and the processed data attribute. Files corresponding to these attributes can be retrieved using links on this web page. The Test 5B data set web page also

provides access to images, videos, webcam images and webcam videos (as well as timelapse videos and composite videos) for Test 5B.

## 5.4. Summary

A typical experimental task from the Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems project is presented as an example of applying the Lehigh RTMD Metadata Web to a typical experiment. A brief introduction of the experimental task was given. The application of the Lehigh RTMD Metadata Web was presented in figures that show the information and data related to the experiment. Information about the project and the typical experimental task is represented on web pages at the project class level and the experimental task class level. The configuration of a typical experimental setup is represented on the web pages at the test condition class level, with detailed information about the specimen, facility, actuators, loading fixtures, bracing and reaction fixtures, sensors, and cables. The loading protocol for the tests is represented on the web page at the test class level. Experimental data, such as raw data, corrected data, processed data, and images and videos from cameras and webcams are represented on web pages at the data set class level.



a. Prototype Structure



c. Photo of Test Setup

Figure 5.1 Experiments on PT Connection with BFFD (Adopted from Wolski, 2006)

		UNIVERSITY.	
SELF-CENTERING DAMAGE-I	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]	
Projects	Project: Self-Centering Damage-Fr Steel Frame Systems	ee Seismic-Resistant	
🔍 Search (beta)	Name:         Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems           Description:         The project is investigating a family of innovative self-centering (SC) stee frame systems with the potential to withstand the currently accepted design basis earthquake (DBE) for buildings wi [more]		
	Descriptive Files     Descriptive Files     Presentations	tions	
	▼ Typical Experimental Tasks ③ Ta[Typical Experimental Tasks] Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems ↓	Experimental Tasks O s Tasks O	
	Edit		
	LEHIGH UNIVERSIT	Y RTMD METADATA WEB (VERSION 1.2)	

Figure 5.2 Self-Centering Damage Free Seismic-Resistant Steel Frame Systems Project Web Page
SELF-CENTERING DAMAGE-FRI	E SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]	
Projects	Typical Experimental Task: Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems         Name:       Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems         Description:       This experimental task evaluates the development of an energy dissipati (ED) device for earthquake resistant self-centering steel moment resisti frames (SC-MRFs) with post-tensioned steel moment c [more]		
	V Descriptive Files	Publications     4ICEE conference paper     M. Wolski Thesis     BFFD Status Report 1	
	▼ Presentations Presentation by M. Wolski ‡		
	▼ Test Condition          ⓐ Test Condition 0 \$ Test Condition 1 \$ Test Condition 2a \$ Test Condition 3 \$ Test Condition 3 \$ Test Condition 4 \$ Test Condition 6 \$ Test Condition 6 \$ Edit X Delete	Analyses ③	
	🖙 Back To Project		

Figure 5.3 Task 4: Develop Energy Dissipation Elements Appropriate for SC-MRF and SC-CBF Systems Web Page



(a) BFFD Status Report 1 Nov 2005.pdf

(b) M. Wolski Thesis.pdf

\*



(c) 4ICEE conference paper No.108.pdf

Figure 5.4 Files for Publications

		EEHIGH UN I VERSITY.	
SELF-CENTERING DAMAGE-FREE	SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]	
Projects	Test Condition: Test Condit	tion 5	
🔍 Search (beta)	Name:         Test Condition 5           Description:         Test Condition 5 was used to evaluate the performance of the BFFD with the design friction force under EQ loading and also the evaluate the effect of bearing.           Logarithm         Test A, B, & C were reported as [more]		
	Descriptive Files	Setup Drawings and Photos ③	
		setup drawings.dwg 🎗	
	▼ Persons ③	▼ Specimen	
	Richard Sause ¥ × James M. Ricles \$ × Michael F. Wolski \$ ×	Specimen #2 ×	
	Thomas M. Marullo 🇘 🗙	Facility     Peoletime Multi-Directional (RTMD)	
	Peter Bryan 🎚 🗙	Facility ×	
	▼ Actuators ③	Other Loading Fixtures	
	NEES Actuator 2 🎗 🗙	Actuator clevis plates 🗘 🗙	
	Bracing and Reaction Fixtu      Reaction fixture for actuator      X     Reaction fixture for specimen      X     Bracing fixture for specimen (adjusted)     X	▼ Sensors LVDT-1	
		Strain Gage (SG-15 B) 🖡 🗙 Strain Gage (SG-15 C) 🏶 🗙	
	Cables		
	▼ Tests O		
	Test 5A I Test 5B I		
	Test 5C \$		
	🛃 Edit		
	× Delete		
	🖙 Back To Typical Experimental Task		
	LEHIG	H UNIVERSITY RTMD METADATA WEB (VERSION 1.2)	

Figure 5.5 Test Condition 5 Web Page

		EFHIGH
SELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Specimen: Specimen #2	
🔍 Search (beta)	Name: Specimen #2 Description: An exterior PT connection with a BFFD was section was a W36x300, which was scale specimen. The scaled beam was a A572 Gra	investigated. The prototype beam d by a factor of 3/5 in the test de 50 W21x111 <u>(more)</u>
	▼ Specimen Drawings and Phot setup drawings.dwg t	
	▼ Specimen Components Beam ↓ Column ↓ BFFD (Fillet Weld) ↓ PT Strands ↓ Top Flange Slotted Keeper Angle ↓	
	Managed By: Self-Centering Damage-Free Seismic-Resist	ant Steel Frame Systems
	Back To Test Condition	
	LEHIGH UNIVERSITY I	RTMD METADATA WEB (VERSION 1.2)

Figure 5.6 Specimen # 2 Web Page



Figure 5.7 Tab E-1 of Setup Drawing File

SELF-CENTERING DAMAGE	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout] ED (Fillet \\(ald)
Search (beta)	Name: BFFD (Fillet Weld) Description: The scaled BFFD consisted of welded to the beam with a 9 [more]	f a slotted plate made from a 1″x1′x1′x0′%″ plate 9/16 inch fillet weld. The built-up column angles
	▼ Geometry Geometry of BFFD	Location Using Drawings an Location of BFFD
	Location Using Grid      Steel Materials	Concrete Materials
	Material of Bolts 1" dia, \$ × Material of Bolts 5/8" dia, \$ × Material of Column Angle Long Leg \$ × Material of Column Angle Short Leg \$ ×	Rebar Materials     Other Materials
	Material of Slot Plate 🗘 🗙	Material of Friction Plate 🗘 🗙
	K Delete     Gack To Specimen	

Figure 5.8 BFFD (Fillet Weld) Web Page

		UNIVERSITY.
SELF-CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Location Using Drawings and Photos	s: Location of BFFD
Search (beta)	Name: Location of BFFD Description: Location of Beam related with other sp Drawings.dwg", under Tab E-1.	ecimen can be found in "Setup
	setup drawings.dwg	
	Celt Celte Component Component	
	LEHIGH UNIVERSITY	RTMD METADATA WEB (VERSION 1.2)

Figure 5.9 Location of BFFD Web Page

SELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Geometry: Geometry of BFFD	
🔍 Search (beta)	Name: Geometry of BFFD Description: The drawing of its geometry can be foun- tabs S-1.	d in "Setup Drawings.dwg", under
	Drawings and Photos     Setup drawings.dwg \$	
	i Edit ■ Belete	
	Component	

Figure 5.10 Geometry of BFFD Web Page



Figure 5.11 Tab S-1 of Setup Drawing File

		EEHIGH
SELF-CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Steel: Material of Slot Plat	te
🔍 Search (beta)	Name: Material of Slot Plate Description: Type of Steel: A572 Grade 50	
	▼ Modulus of Elasticity	▼ Yield Stress
	Modulus of Elasticity	Yield Stress
	▼ Ultimate Stress	► Material Property Files ③
	Ultimate Stress	
	Managed By: Self-Centering Damage-Fre	e Seismic-Resistant Steel Frame Systems
	LEHI	GH UNIVERSITY RTMD METADATA WEB (VERSION 1.2)

Figure 5.12 Material of Slot Plate Web Page

SELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Yield Stress: Yield Stress	
Search (beta)	Name: Yield Stress Nominal Value: 50 ksi Average Value: 66.9 ksi Test Values C C C C C C C C C C C C C	

Figure 5.13 Yield Stress Web Page

		UNIVERSITY.
SELF-CENTERING DAMAGE-FR	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Other Material: Material of Fricti	on Plate
Search (beta)	Name: Material of Friction Plate Description: UNS 260 half-hard cartridge brass	
	Material Property Files	
	Managed By: Self-Centering Damage-Free Seismic-F	Resistant Steel Frame Systems
	LEHIGH UNIVER	SITY RTMD METADATA WEB (VERSION 1.2)

Figure 5.14 Material of Friction Plate Web Page

ELF-CENTERING DAMAGE	-FREE SEISMIC-RESISTANT	steel FRAME SYSTEMS	chc305@lehigh.edu [logo Directional (RTMD) Facility
Search (beta)	Name: Description: Origin of Coordinate System:	Real-time Multi-Direction The Lehigh NEES Equination testing using method, or the (more)	onal (RTMD) Facility ipment Site will have the capabilities to perfi- ng the effective force method, pseudo-dyna e pseudo-dynamic hybrid testing method for
	▼ Site		▼ Specifications ③
	ATLSS Research University) ×	Center (Lehigh	Servo-Controller Specification.doc IT Architecture Specification.doc Data Acquisition System Sepcification.doc
	Facility Drawing	ngs and Phot 🧕	
	<ul> <li>Facility Drawi</li> <li>Reference Pla</li> </ul>	ngs and Phot G	<ul> <li>Reference Planes</li> </ul>
	Facility Drawi      Reference Pla      Reference Pla	nes for React	Reference Planes
	Facility Drawi      Reference Pla      Reference Pla      Managed By: RTI	Ine for Strong ( Ines for React ( MD Site Spec	Reference Planes

Figure 5.15 Real-Time Multi-Directional (RTMD) Facility Web Page





(c) Data Acquisition System Sepcification.doc

Figure 5.16 Files for Facility Specifications

AS MURTI-DIRECTIONAL TESTING FACILITY		×
ELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logoւ
	Name: NEES Actuator 2 Description:	
Search (Deta)	✓ Descriptive Files     ③     Servotest Actuator 2 Calibration     Certificates Q1-07.pdf      Unification	▼ Specifications ③ Actuator 2.doc \$
		Drawings and Photos     Actuator Clevis.pdf
	► Fixed Node (AFN) Location ③	Fixed Node (AFN) Location ③
	Structure Node (ASN) Locat ③	Structure Node (ASN) Locat ③
	Managed By: RTMD Site Spec	
	🖙 Back To Test Condition	

Figure 5.17 NEES Actuator 2 Web Page





Here we 5 hybrids: stratures. Two of dense have a semisorm lood reporting of VDMX is resonance to the strature large strature large maximum lood reporting of the forward in the strature large strature

HYDRAULIC ACTUATORS

(b) Hydraulic Actuators.pdf



(c) Actuator 2.doc



(d) Actuator Clevis.pdf

Figure 5.18 Descriptive, Specification, and Drawing Files for NEES Actuator  $\mathbf{2}$ 

		UNIVERSITY.
SELF-CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Bracing and Reaction Fixture: Brac specimen (adjusted)	cing fixture for
Search (beta)	Name: Bracing fixture for specimen (adjusted) Description: Four W12x190 columns were placed ve setup. W21x101 spreader beams we spanning over the W14x426 column at <u>Imore</u> ] <b>Fixture Drawings and Photo</b> setup drawings.dwg <b>t</b> Managed By: Self-Centering Damage-Free Seismic-Res Edit Content of the sector	ertically at the corners of the test re connected to these columns, each end, with high-strength g
	LEHIGH UNIVERSIT	Y RTMD METADATA WEB (VERSION 1.2)

Figure 5.19 Bracing Fixture for Specimen (Adjusted) Web Page

			I u	LEHIGH N I V E R S I T Y
SELF-CENTERING DAMAGE-	FREE SEISMIC-RESISTANT ST	EEL FRAME SYSTEMS	chc305@	lehigh.edu [logout]
Projects	Sensor: LVD	T-3		
Search (beta)	Name: Description: Calibration Constar Calibration Date:	LVDT-3 To measure the de side. nt: Sep 07 2005 9:09	eflection at flange of column spec	imen at east
	▼ Calibration LVDT 3.pdf		▼ Specifications LVDT.doc \$	٢
	Location Using     Location of LVDT-3	) Drawings an	► Location Using Grid	٢
	Managed By: Self-C	Centering Damage-Fre	e Seismic-Resistant Steel Frame	Systems
	🗙 D elete			
	🖙 Back To Test Co	ondition		
		LEHI	GH UNIVERSITY RTMD METADATA	NEB (VERSION 1.2)

Figure 5.20 LVDT-3 Web Page





RTAD		EHIGH
SELF-CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Name: Location of LVDT-3	s: Location of LVDT-3
🔍 Search (beta)	Description: Precise location for Displacement Transduct Drawings.dwg", under tabs IN-2.	er can be found in "Setup
	Drawings and Photos     Setup drawings.dwg	
	Content of the second of the	
	LEHIGH UNIVERSITY	RTMD METADATA WEB (VERSION 1.2)

Figure 5.22 Location of LVDT-3 Web Page



Figure 5.23 Tab IN-2 of Setup Drawing File

		EHIGH
SELF-CENTERING DAMAGE-FRE	E SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logout]
Projects	Cable: Cables          Name:       Cables         Description: In all Test Condition, except for Test Cond. long) and 3 strain cables (8 channel, 50' lor         Cable ID:         Managed By: Self-Centering Damage-Free Seismic-Resis         Edit         X Delete         ← Back To Test Condition	6, 6 voltage cables (4 channel, 50' ng) were used.
-	LEHIGH UNIVERSITY	RTMD METADATA WEB (VERSION 1.2)
	E' 504011 WID	

Figure 5.24 Cables Web Page

				UNIVERSITY.
SELF-CENTERING DAMAGE-FREE	SEISMIC-RESISTANT	STEEL FRAME SYSTEMS		chc305@lehigh.edu [logout]
Projects	Test: Test !	5B		
Search (beta)	Name: Description:	Test 5B Test 5B consisted of a rate that was 8 times s +0.0245 and -0.02 radia	earthquake loading v lower than real time. ns. The nominal initia	which was conducted at a This test had a ?r,max of I PT stran <u>[more]</u>
	Start Date and Time:	Jan 30 2006 11:03 am		
	End Date and Time:	Jan 30 2006 11:21 am		
	Uncontrolled Tes Condition:	t		
	Descriptive Fi	les 📀		
	▼ Test Protoco	bl	▼ Persons	۲
	Test protocol #1	4 ×	Michael E. Wolski 🗘 Thomas M. Marullo Peter Bryan 🗘 🗙	x I x
	Data Set     Test 5B data set			
	🛃 Edit			
	🗙 D elete			
	🖙 Back To Test	Condition		
		LEHIG	UNIVERSITY RTMD ME	TADATA WEB (VERSION 1.2)

Figure 5.25 Test 5B Web Page

ELF-CENTERING DAMAGE	FREE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [logou
Projects	Test Protocol: Test protoc	ol #14
Search (beta)	Name: Test protocol #14 Description: Loading protocol is EQ load 0.0245.	ling with frequency 0.05 and qr max (rads) is
	► Descriptive Files	
	▼ History File	<ul> <li>Simulation Method Files</li> </ul>
	historyFile.csv	MikeWDispTest.m \$ sim.xml \$
	▼ DAQ Config Files	Controller Config Files
	daq.xml 🎗	► Other Config Files ③
	🛃 Edit	
	🗙 D elete	
	🖛 Back To Test	

Figure 5.26 Test Protocol # 14 Web Page

CENTERING DAMAGE-F	REE SEISMIC-RESISTANT STEEL FRAME SYSTEMS	chc305@lehigh.edu [k
5	Data Set: Test 5B data s	et
🔍 Search (beta)	Name: Test 5B data set Description: An FFT was performed o transform computes the p The data was then filtered	n the actuator data (load & displacement) ower spectral density (psd) of the data. I using SigmaPlot with a Butterw [more]
	▼ Descriptive Files	Raw Binary Data
	5B (Test 4) Processed Data.txt \$	▼ Raw Engineering Unit Data 《 DAQ 5B.txt \$ All Data.csv \$
	▼ Corrected Data	Processed Data
	All Data dec 52.csv Actuator Data filtered.JNB	5-B (Test 4) Processed.xls ‡
	Actuator Data filtered.CSV + Actuator Data filtered dec 52.csv ‡	▼ Images C images.zip \$
	► Videos	🗊 🔻 Webcam Images
		webcams.zip ‡
	▼ Timelapsed Video	Composite Video
	cam12.m1v ↓ cam11.m1v ↓	5BComposite.m1v 🎗
	cam10.m1v ↓	► Other Files
	🛃 Edit	
	🗶 D elete	
	🖛 Back To Test	

Figure 5.27 Test 5B Data Set Web Page

## Chapter 6

# Example of Lehigh RTMD Metadata Web Applied to Hybrid Simulation

### 6.1. Introduction

A project, named the Studies on Large-scale, Real-time Pesudodynamic (PSD) Testing (Mercan, 2007), is used as an example of applying the Lehigh RTMD Metadata Web to a hybrid simulation. The hybrid simulations of this project were conducted at the RTMD earthquake simulation facility. The information and data associated with the selected simulation are organized using the Lehigh RTMD Metadata Web.

## **6.2. Background of Project**

The Studies on Large-scale, Real-time Pseudodynamic (PSD) Testing project (Mercan, 2007) was selected as an example of applying the Lehigh RTMD Metadata Web to a hybrid simulation. The research objectives of this project are to develop a methodology for implementing large-scale, real-time hybrid PSD simulation, performing experiments, and evaluating the simulation results. The hybrid PSD simulation is a displacement-based experimental technique that is used to simulate the seismic response of structures. It utilizes feedback signals from a test structure in a numerical integration algorithm to sequentially solve the equations of motion to determine command displacements. The command displacements are imposed on the test structure using hydraulic actuators. It is advantageous to avoid the cost of fabricating and testing the entire structure. The hybrid PSD simulation includes only a portion of the structure in the laboratory.

The prototype building for the hybrid PSD simulation is a 3 story MDOF shear building with an elastomeric damper in the first story as shown in Figure 6.1. The 3 story MDOF frame is modeled in the Simulink 6.2 as an analytical substructure for the simulation. The elastomeric damper is the physical substructure of the simulation, located in the RTMD earthquake simulation facility. The drawings and photos of the physical substructure setup are presented in Figure 6.2.

A Simulink model is used for the real-time hybrid PSD simulations as shown in Figure 6.3. SUBSYSTEM 1 is the model for the integration algorithm (as well as simulation coordinator in the Lehigh Model) and the analytical substructure. The physical substructure is SUBSYSTEM 2. SUBSYSTEM 2 receives command from the integration algorithm in SUBSYSTEM 1 to control the servo-hydraulic system in the laboratory (which imposes displacement on this physical substructure) and then sends feedback from the physical substructure to SUBSYSTEM 1. SUBSYSTEM 1 then performs a state determination for the analytical substructure. The data from the physical and analytical substructures are combined and the integration algorithm computes the next command/displacement.

#### 6.3. Information and Data in Lehigh RTMD Metadata Web

The data and information related to the selected project has been organized using the Lehigh RTMD Metadata Web. For simplicity, only selected web pages and related files are presented for this application example. The information and data are presented in the five main levels of the Lehigh Model.

### 6.3.1. Project Class Level and Experimental Task Class Levels

The information about the project and the hybrid experimental task is presented in the web pages shown in the Figures 6.4 and 6.5, which represent objects of the project class and the hybrid experimental task class of the Lehigh Model, respectively. In the Studies on Large-scale, Real-time Pseudodynamic (PSD) Testing project web page, a concise description about the project and its research objectives is given. A file containing the researcher's dissertation shown in Figure 6.6a can be accessed from the publications attribute to provide more detailed information regarding this project. Figure 6.4 shows that the project contains two typical experimental tasks and two hybrid experimental tasks, the Hybrid PSD Test with MDOF Analytical Substructure Task is the selected example of an application of the Lehigh RTMD Metadata Web to a hybrid simulation. In the Hybrid PSD Test with MDOF Analytical Substructure Task web page shown in Figure 5.5, the description briefly summarizes the test method used for this task. A conference paper related to this hybrid experimental task shown in Figure 6.6b can be accessed from the publications attributes. Only one simulation coordinator, one analytical substructure, and one physical substructure are used for this hybrid experimental task as shown in Figure 6.5.

#### 6.3.2. Test Condition Class Level

The Target PC web page shown in Figure 6.7 represents an object of the simulation coordinator class of the Lehigh Model. The description and descriptive files provide detailed information about the simulation coordinator. The software used to conduct this experiment is called xPC, as shown in Figure 6.7. The object link in the model

attribute provides access to the Integration Algorithm web page shown in Figure 6.8 to represent information about the model used for this simulation coordinator. A description of the model is given. The model file can be accessed from the model file attribute. As described in Section 6.2, the simulation coordinator and the analytical substructure are contained in one model for this hybrid PSD simulation. A file in the visual files attribute indicates that SUBSYSTEM 3 in the model is used for the simulation coordinator as shown in Figure 6.9. The simulation coordinator for this hybrid PSD simulation is located in the Real-Time Multi-Directional (RTMD) facility, as shown in Figure 6.7. The information about this facility is the same as described in the example in Chapter 5.

The 3 Story Substructure web page shown in Figure 3.10 represents an object of the analytical substructure class of the Lehigh Model. The description and descriptive files provide information about this analytical substructure. Simulink 6.2 is the software used to model the analytical substructure for this hybrid simulation. The information about the analytical substructure model is represented in the 3 DOF frame web page shown in Figure 6.11. The description briefly describes the model. The model file can be accessed from the model files attribute. A file is provided through the visual files attribute to indicate that Embedded Function 1 in the model is used for the analytical substructure as shown in Figure 6.12. The analytical substructure for this hybrid PSD simulation is located in the Real-Time Multi-Directional (RTMD) facility as shown in Figure 6.10.

The Damper Testbed web page in Figure 6.13 represents an object of the physical substructure class of the Lehigh Model. A description of this physical substructure is given. An AutoCAD file in the setup drawings and photos attributes provide an illustration of the physical substructure setup in the laboratory as shown in Figure 6.2a. Detailed information about the Damper Specimen is represented in the Damper Specimen web page shown in Figure 6.14. The web page includes a description, and drawings and photos for the specimen. The value of the facility attribute indicates that this physical substructure is located in the Real-Time Multi-Directional (RTMD) facility as shown in shown in Figure 6.13. NEES Actuator 2 is used for this physical substructure. The information about the bracing and reaction fixtures and other loading fixtures used for this physical substructure setup are presented in Figures 6.15 and 6.16.

#### 6.3.3. Test Class Level

The information about a simulation and related analytical substructure computations and physical substructure tests is represented in the web pages shown in Figures 6.17 to 6.19, which represent objects of the simulation class, the analytical substructure computation class, and the physical substructure test class of the Lehigh Model, respectively. The MDOF (KFF=2) Simulation web page shown in Figure 6.15 contains information about the simulation and the relationship among related analytical substructure computations and physical substructure tests. The object link in the interfaces to analytical substructure computations attribute provides access to the Linear MDOF 3 Story Substructure web page shown in Figure 6.20, which represents an object of the analytical

sub. interface class of the Lehigh Model. This web page contains the command and feedback signals between the simulation and the related analytical substructure computation. The information about the analytical substructure computation can be accessed from the computation attribute, which links to the web page shown in Figure The Linear MDOF (KFF=2) Computation web page (in Figure 6.18) also links to 6.18. the Linear MDOF 3 Story Substructure web page (in Figure 6.20) from the link in the interface to simulation attribute. Similarly, the link in the interfaces to physical substructure tests attribute (in Figure 6.17) provides access to the Linear MDOF Damper Testbed web page shown in Figure 6.21. This page contains the command and feedback signals between the simulation and the related physical substructure test. The information about the physical substructure test can be accessed from the physical substructure test attribute, which links to the web page shown in Figure 6.19. The Linear MDOF (KFF=2) Test web page (in Figure 6.19) also links to the Linear MDOF Damper Testbed web page (in Figure 6.21).

#### 6.3.4. Test Data Set Class Level

For each hybrid simulation, the experimental data is stored in the web pages shown in Figures 6.22 to 6.24. The web pages shown in Figures 6.22 and 6.23 represent objects from the analytical data set class of the Lehigh Model. The Data Set web page for the simulation (in Figure 6.22) includes the output files from the hybrid simulation. The Data Set web page for the computation (in Figure 6.23) includes the output files from the analytical substructure computation. The Data Set web page for the physical test shown in Figure 6.24 represents an object of the experimental data set class of the

Lehigh Model. This web page contains the experimental data, such as raw data, images, and videos, from the associated physical substructure test.

## 6.4. Summary

A hybrid simulation is used as an example of applying the Lehigh RTMD Metadata Web. The selected project was the Studies on Large-scale, Real-time Pseudodynamic (PSD) Testing project. The Lehigh RTMD Metadata Web example was presented using figures showing the information and data related to the selected simulation. Information on the project and a typical experimental task was shown on web pages at the project class level and the experimental task class level. The simulation coordinator, analytical substructure, and physical substructure setups are represented in web pages at the test condition class level. The simulations with the related analytical substructure computations and the related physical substructure tests are presented on web pages at the test class level. The relationships between the simulation and the related analytical substructure computations, and the simulation and the related physical substructure tests are also presented at the test class level. The resulting data, such as raw data, processed data, and images and videos and output files, are presented on web pages at the data set level.



Substructure





a. Drawing of Physical Substructure Setup





b. Photo of Physical Substructure Setup Figure 6.2 Physical Substructure Setup (Adopted from Mercan, 2007)



Analytical Substructure and an Elastomeric Damper as Physical Substructure (Adopted from Mercan, 2007)

STUDIES ON LARGE-SCALE	, REAL-TIME PSEUDODYNAMIC TESTING		chc305@lehigh.edu [logout
Projects	Project: Studies on La Pseudodynamic Testing	rge-Scale, R 9	eal-time
🔍 Search (beta)	Name: Studies on Large-Scali Description: The pseudodynamic (F technique that is use utilizes feedback signa	e, Real-time Pseudo PSD) test method is of to simulate the als from a test struc	dynamic Testing a displacement-based experimenta seismic response of structures. I ture in a [more]
	► Descriptive Files	O Public	ations 💿
		Dissertat	ion_Oya.pdf 🎗
	▶ Presentations	0	
	Typical Experimental Ta	O Thybrid	l Experimental Tas 💿
	Predefined Sinusoidal Displacemen	t Hybrid Ps	SD Tests with MDOF Analytical
	PSD Method of the Elastomeric Dar Task	npers Hybrid PS Substruct	D Tests with SDOF Analytical ture Task 🗣
	► Analysis Tasks	0	
	🛃 Edit		

Figure 6.4 Studies on Large-Scale, Real-Time Pseudodynamic Testing Web Page

STUDIES ON LARGE-SCALE,	REAL-TIME PSEUDODYNAMIC TESTING		chc305@lehigh.edu [logout]
Projects	Hybrid Experimental Analytical Substructu	Fask: re Tas	Hybrid PSD Tests with MDOF sk
🔍 Search (beta)	Name: Hybrid PSD Tests wit Description: Hybrid pseudodynam	n MDOF 4 ic test us	Analytical Substructure Task sing Alpha integration method.
	► Descriptive Files		▼ Publications ③
			Structures Congress 2007.doc ‡
	▶ Presentations	٢	
	Simulation Coordinator		Analytical Substructure
	Target PC		3 Story Substructure I
	Physical Substructures	٢	► Analyses
	Damper Testbed 🎗		
	<b>Edit</b>		
	× Delete		
	🖛 Back To Project		

Figure 6.5 Hybrid PSD Tests with MDOF Analytical Substructure Task Web Page

And start and Encoderated Station on Long State
Real-time Pseudodynamic Testing
by
Oya Mercan
Presented to the Graduate and Research Committee
of Lehigh University
in Candidacy for the Degree of
Doctor of Philosophy
in
Department of Civil and Environmental Environmental
Department of Civit and Environmental Engineering
Lehigh University
May, 2007

# a. Dissertation\_Oya.pdf

# Implementation of Real-time Hybrid Pseudodynamic Test Method for Evaluating Seismic Hazard Mitigation Measures

#### Authors:

Oya Mercan, Lehigh University, 117 ATLSS Drive, ATLSS Center, Bethlehem, PA 18015, oym2@lehigh.edu James M, Ricles, Lehigh University, 117 ATLSS Drive, ATLSS Center, Bethlehem, PA 18015, jmz?@lehigh.edu

#### ABSTRACT

Real-time hybrid pseudodynamic (PSD) testing of a 3-story steel moment resisting frame (MRD) requiped with elastometric dampers at the first floor is presented. The components of the serve-hydraulic integrated control system of the NEES Real-time Multi-directional (RTMD) Enthquake Simulation Facility at Leagh University are described, along with the analytical substructure modeling, experimental substructure properties and the experimental substructure modeling experimental substructure, properties and desprintental substructure entropy in evaluated. In order to minimize the elay in the measured restoring force from the experimental substructure, a velocity feed forward component is introduced into the serve-control loop and its contribution is demonstrated both analytically and experimentally. Selected experimental substructure, a velocity feed formal component is presented.

#### INTRODUCTION

INTRODUCTION Introducing properly designed sopplemental passive damping devices is an effective means for streams hazard mingation of structures. Daring a seismic event, structural damage can be alleviated by dissipating energy in damping devices. As a result, the lateral darft, the base these, overtuning moment and damage to nonstructural elements are reduced. To develop an understanding of the effectiveness of the use of these devices would be effective to the set of the effectiveness of the use of these devices ownards reducing sensing change in structures, testing of the devices in conjunction with the structural system are required. Because these devices are typically sensitive to the rate of loading, the testing must be provided a visible afficient of dynamic behavior of complex structures. Fighend testing combines physical testing with numerical simulation [Dermitzakis and Mahin, 1959], and provides a visible afficient response of the structure is calculated from the integration of the equations of motion. The displacements we simulate is calculated from the integration of the equations of motion. The displacements are simulationally indexed to the test structure using fortune calculates of motion.

b. Structure Congress 2007.doc

Figure 6.6 Files for Publications

		UNIVERSITY.
STUDIES ON LARGE-SCALE, REA	AL-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logout]
Projects	Simulation Coordinator: Ta	rget PC
Search (beta)	Name: Target PC Description: The overall structure (simul (analytical susbstructure) and relationship among them is pro Software: xPC	laiton coordinator) includes a 3 DOF frame a Damper Testbed (physical substructure). The esent in MOD Shear Build [ <u>more</u> ]
	Descriptive Files     MDOF Shear Building.jpg	
	Model     Integration Algorithm	▼ Persons ② James M. Ricles
	▼ Facility Real-time Multi-Directional (RTMD) Facility ×	
	▼ Simulations Linear MDOF (KFF=0) Simulation \$ Linear MDOF (KFF=2) Simulation \$ Non-Linear MDOF (KFF=0) Simulation \$ Non-Linear MDOF (KFF=2) Simulation \$	
	☑ Edit ★ Delete	
	🖙 Back To Hybrid Experimental Task	

Figure 6.7 Target PC Web Page

STUDIES ON LARGE-SCALE	, REAL-TIME PSEUDODYNAMIC TESTING		chc305	i@lehigh.edu [logout
rojects	Model: Integration Al	gorith	וm	
🔍 Search (beta)	Name: Integration Algorithm Description: The model file include coordinator, the sime Algorithm (model file	s both a lation co .jpg in d	analystical substructure and s oordinator is the highlighted a lescriptive files.	imulation area of Intergratio
	▼ Model Files	٢	▼ Visual Files	٢
	SimulationCoordinatorModel.mdl	ţ	Intergration Algorithm (mo	odel file).jpg 🎗
	🛃 Edit			
	× Delete			
	🖛 Back To Simulation Coordinato	r		

Figure 6.8 Integration Algorithm Web Page



Figure 6.9 Subsystem 1 of Simulink Model for Real-Time Hybrid PSD Simulation – Simulation Coordinator in Highlighted Box

		UNIVERSI
UDIES ON LARGE-SCALE, pjects	Analytical Substructure: 3 Story S	chc305@lehigh.edu [logo Substructure
🔍 Search (beta)	Name: 3 Story Substructure Description: The model simulates real-time hybrid PS building (shown as analytical substruc decriptive files) with connection to the ex Software: Simulink 6.2	D test of the three-story linear sho ture in MODF Shear Building.jpg xperi [ <u>more]</u>
	▼ Descriptive Files	
	MDOF Shear Building.jpg <sup>‡</sup> MDOF Shear Building Properties.jpg <sup>‡</sup>	
	▼ Model 3 DOF frame James M Oya Mer Thomas	ns. I. Ricles I × rcan I × M. Marullo I ×
	▼ Facility Real-time Multi-Directional (RTMD) Facility ×	
	▼ Computations ③	
	Linear MDOF (KFF=0) Computation Linear MDOF (KFF=2) Computation Non-Linear MDOF (KFF=0) Computation Non-Linear MDOF (KFF=2) Computation	
	₽	
	× Delete	
	🖛 Back To Hybrid Experimental Task	

Figure 6.10 3 Story Substructure Web Page

		sha205@lakish.adu.Daasu	
Projects	Model: 3 DOF frame	chessistenigineau fugua	
	Name: 3 DOF frame Description: The model file includes both analystical substructure and simulation coordinator, the analytical substructure is the highlighted area of 3 DOF frame (model file).jpg in descriptive files.		
	Model Files     SimulationCoordinatorModel.mdl	▼ Visual Files ③ 3 DOF frame (model file).jpg \$	
	Edit X Delete		
	🖙 Back To Analytical Substructure		

Figure 6.11 3 DOF Frame Web Page



Figure 6.12 Subsystem 1 of Simulink Model for Real-Time Hybrid PSD Simulation – 3 DOF Frame Analytical Substructure in Highlighted Box

STUDIES ON LARGE-SCALE, REA	L-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logout]
Projects	Physical Substructure: Dar	nper Testbed
🔍 Search (beta)	Name: Damper Testbed Description: Two existing A-frames were A-Frame) was connected to 1 the larger A-frame (North A-F	used in the setup. The smaller A-frame (South he actuator at the south end of the setup, while rame) supported the [more]
	Descriptive Files	▼ Setup Drawings and Phot ③ Test Setup.dwg \$
	▼ Persons ③	▼ Specimen
	James M. Ricles 🖡 🗙 Oya Mercan 🖡 🗙	Damper Specimen 🗙
	Thomas M. Marullo 🖡 🗙	▼ Facility Real-time Multi-Directional (RTMD) Facility ×
	• Actuators	• Other Loading Eixtures
	NEES Actuator 2 🖡 🗙	Clevis Plates 🖡 🗙 Welded T 🖡 🗙
	Bracing and Reaction Fi	▼ Sensors
	North A-Frame 1 ×	
	South A-Frame <b>*</b> × Extension Column <b>*</b> ×	TDT-A I ×
	Lateral Tie Bar 🎙 🗙 W12X133 Column Section 🇘 🗙 Roller Bearings 🎝 🗙	► Cables ③
	Roller Bearings V ×   Physical Substructure T  Linear MDOF (KFF=0) Test  Linear MDOF (KFF=2) Test  Non-Linear MDOF (KFF=2) Test  Non-Linear MDOF (KFF=2) Test  Edit  Celt Celt Celt Celt Celt Celt Celt Ce	H UNIVERSITY RTMD METADATA WEB (VERSION 1.2)

Figure 6.13 Damper Testbed Web Page

		ELEHIGH
STUDIES ON LARGE-SCALE,	REAL-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logout]
Projects	Specimen: Damper Specimen	
🔍 Search (beta)	Name:         Damper Specimen           Description:         Three, 254 mm (10 in) long, steel tubes tube contains a longitudinal steel bar, 55 section of 25.4 mm (1 in) x 50.8 mm (2 in)	are welded together and each steel 33 mm (21 in) in length, with a cross ) wra [ <u>more]</u>
	▼ Specimen Drawings and P ③	
	Test Setup.dwg \$ 1.jpg \$ 2.jpg \$ 3.jpg \$ 4.jpg \$ 5.jpg \$ 6.jpg \$ 7.jpg \$ 9.jpg \$ 10.jpg \$ 11.jpg \$ 13.jpg \$	
	▼ Specimen Components Steel Damper Tube ↓ Longitudinal Bar ↓ Bolted Transverse Bar ↓ Transverse Attachment Bar ↓ Elastomer Damper ↓	
	Managed By: Experimental Investigation of a Prototype Elastomeric Structural Damper	
	🛃 Edit	
	× Delete	
	🖙 Back To Physical Substructure	
	LEHIGH UNIVERSIT	Y RTMD METADATA WEB (VERSION 1.2)

Figure 6.14 Damper Specimen Web Page

RTMD	🖲 LEHIGH
PROJECTS >> STUDIES ON LARGE-SCALE, REAL-TIME PSEUDODYNAMIC TESTING	admin [logout]
Bracing and Reaction Fixture: North A-Frame	
⇐ Back To Physical Substructure	
Name: North A-Frame Description: North A-Frame was connected to the damper assembly via the Extension Column, acting as a fixture.	reaction
Texture Drawings and Photos       Test Setup.dwg ‡       A-Frames.dwg ‡	
Managed By: Experimental Investigation of a Prototype Elastomeric Structural Damper	
📓 Edit	
× Delete	
MET	ADATA WEB V1.4

Figure 6.15 North A-Frame Web Page

RTMD				
PROJECTS >> STUDIES ON LARGE-SCALE, REAL-TIME PSEUDODYNAMIC TESTING				
Other Loading Fixture: Clevis Plates	Other Loading Fixture: Clevis Plates			
🖙 Back To Physical Substructure				
Name:         Clevis Plates           Description:         Two 4 inch thick actuator clevis plates were used in this test condition. One is to connect the actuator damper assembly; the other is to connect the actuator and south A-Frame.				
▼ Geometry	Location Using Drawings and Photos	۲		
Geometry of Clevis Plates	Location Using Grid	٢		
▼ Steel Materials ③	Concrete Materials	٢		
steel 🎚 🗙	Rebar Materials	۲		
	Other Materials	٢		
Managed By: Experimental Investigation of a Prototype Elastomeric Structural Damper          Edit         X Delete				
	METAD	ATA WEB V1.4		

Figure 6.16 Clevis Plates Web Pages

STUDIES ON LARGE-SCALE,	REAL-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logout]
Projects	Simulation: Linear MDOF (	KFF=2) Simulation
Search (beta)	Name:         Linear MDOF (KFF=2) Simulation           Description:         The simulation is for linear with EQ loading (1994 No acceleration scaled to DBB           Start Date and Time:         Jun 15 2006 10:00 am	ulation r MDOF overall structure (simulation coordinator) rthridge earthquake Canoga Park ground =).
	End Date and Jun 15 2006 11:00 am Time:	
	► Descriptive Files	
	Versons James M. Ricles X × Oya Mercan X × Thomas M. Marullo X ×	
	▼ Interfaces to Physical ③ Linear MDOF Damper Testbed ↓	▼ Interfaces to Analytica ③ Linear MDOF 3 Story Substructure \$
	▼ Simulation Protocol Linear MDOF EQ Protocol (KFF=2) ×	▼ Data Set
	☑ Edit X Delete	
	🖙 Back To Simulation Coordinator	
	LEHIG	H UNIVERSITY RTMD METADATA WEB (VERSION 1.2)

Figure 6.17 Linear MDOF (KFF=2) Simulation Web Page
		UNIVERSITY.
STUDIES ON LARGE-SCALE, R	EAL-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logout]
Projects	Computation: Linear MDOF (KFF=2)	Computation
Search (beta)	Name: Linear MDOF (KFF=2) Computation   Description: The computation is for linear MDOF analy (1994 Northridge earthquake Canoga P DBE).   Start Date and Time: Jun 15 2006 10:00 am   End Date and Time: Jun 15 2006 11:00 am	tical substructure with EQ loading ark ground acceleration scaled to
	Descriptive Files	
	▼ Persons ③ James M. Rides \$ × Oya Mercan \$ × Thomas M. Marullo \$ ×	
	▼ Interface to Simulation     ▼ Comput.       Linear MDOF 3 Story Substructure ×     Linear MDO	ation Protocol F EQ Protocol (KFF=2) ×
	▼ Data Set	
	☑ Edit ★ Delete	
	🖙 Back To Analytical Substructure	
	LEHIGH UNIVERSITY F	RTMD METADATA WEB (VERSION 1.2)

Figure 6.18 Linear MDOF (KFF=2) Computation Web Page

		EEHIGH
STUDIES ON LARGE-SCALE, REA	L-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logout]
Projects	Physical Substructur	e Test: Linear MDOF (KFF=2) Test
	Name: Lir	near MDOF (KFF=2) Test
🔍 Search (beta)	Start Date and Time: Ju End Date and Time: Ju Uncontrolled Test Condition:	n 15 2006 10:00 am n 15 2006 11:00 am
	Descriptive Files	0
	▼ Persons James M. Rides	
	▼ Interface to Simulation Linear MDOF Damper Testbed ×	Test Protocol Linear MDOF EQ Protocol (KFF=2) ×
	▼ Data Set Data Set	
	Edit	
	Back To Physical Substructure	3
		LEHIGH UNIVERSITY RTMD METADATA WEB (VERSION 1.2)

Figure 6.19 Linear MDOF (KFF=2) Test Web Page

piects		
····	Analytical Substructure In Substructure	nterface: Linear MDOF 3 Story
🔍 Search (beta)	Name: Linear MDOF 3 Story Substructur	re
	▼ Commands	▼ Feedbacks
	AnalyticalCommandsfromCoord.csv	AnalyticalFeedbackstoCoord.csv
	Computation	
	Linear MDOF (KFF=2) Computation ×	
	🛃 Edit	
	× Delete	
	🖙 Back To Simulation	

Figure 6.20 Linear MDOF 3 Story Substructure Web Page

TUDIES ON LARGE-SCALE	, REAL-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logo
ojects	Physical Substructure Inte Testbed	erface: Linear MDOF Damper
🔍 Search (beta)	Name: Linear MDOF Damper Testbed	
	▼ Commands	▼ Feedbacks
	PhysicalCommandsfromCoord.csv	PhysicalFeedbackstoCoord.csv
	▼ Physical Substructure T	
	Linear MDOF (KFF=2) Test ×	
	🛃 Edit	
	× Delete	
	Back To Simulation	

Figure 6.21 Linear MDOF Damper Testbed Web Page

TUDIES ON LARGE-SCALE	, REAL-TIME PSEUDODYNAMIC TESTING	chc305@lehigh.edu [logo
Projects	Data Set: Data Set	
🔍 Search (beta)	Name: Data Set Description:	
	Descriptive Files O	utput Data 💿 :2E-1.DAT 🕽
	Processed Data P	nages
	XPC1E-1.csv \$	ideos 📀
	► Other Files	
	Edit	
	× Delete	
	Back To Simulation	

Figure 6.22 Data Set Web Page for Simulation

TUDIES ON LARGE-SCALE	, REAL-TIME PSEUDODYNAMIC TESTING		chc305@lehig	h.edu [logo
ojects	Data Set: Data Set			
Search (beta)	Name: Data Set Description:			
	► Descriptive Files	٢	▼ Output Data	٥
			AnalyticalSubstructure.csv 🎗	
	▶ Processed Data	٢	▶ Images	٢
	► Videos	٢	▶ Other Files	0
	Z Edit			
	× Delete			
	🖛 Back To Computation			

Figure 6.23 Data Set Web Page for Computation

TUDIES ON LARGE-SCALE	, REAL-TIME PSEUDODYNAMIC TESTING	chc305@lel	nigh.edu [logou
ojects	Data Set: Data Set		
🔍 Search (beta)	Name: Data Set Description:		
	► Descriptive Files	▶ Raw Binary Data	0
	Raw Engineering Unit Da 📀	Corrected Data	٢
	Actuator.csv 🎗	► Processed Data	٢
		▼ Images	٢
		DSC_0072.jpg	
	▼ Videos 📀	▶ Webcam Images	0
	Shorten Damper Movie_2.wmv 🎗	► Timelapsed Video	٢
		► Composite Video	٢
	► Other Files		
	<b>E</b> dit		
	× Delete		
	🖛 Back To Test		

Figure 6.24 Data Set Web Page for Physical Test

## Chapter 7

### Summary, Conclusions, and Future Work

#### 7.1 Summary

This research further advanced the Lehigh Model, which was developed previously by Lee et al. (2006), and presented an implementation of the Lehigh Model, the Lehigh RTMD Metadata Web, developed by Marullo (2007). The presentation of the Lehigh RTMD Metadata Web showed how it was developed from the updated Lehigh Model.

The Lehigh Model is a data model for large-scale structural experiments, which was developed at the Real-Time Multi-Directional (RTMD) earthquake simulation facility in the ATLSS Center. The development of the Lehigh Model was based on a review of previous work on data models and databases for structural tests, and a study of large-scale structural experiments conducted at the RTMD earthquake simulation facility. Some of the classes and attributes of the Lehigh Model were updated in this thesis based on further studies of large-scale structural experiments and analytical studies.

A web application, named the Lehigh RTMD Metadata Web, was developed by Marullo (2007) to implement the Lehigh Model. This web application enables users to access and retrieve information and data through the internet. Two examples of applying the Lehigh RTMD Metadata Web were presented in this thesis. First, the Self-Centering Damage-Free Seismic-Resistant Steel Frame Systems Project is an example involving a typical

experiment. Second, the Studies on Large-scale, Real-time Pseudodynamic (PSD) Testing Project is an example involving hybrid simulations.

#### 7.2 Conclusions

The Lehigh Model has been successfully developed to represent the information and data associated with large-scale structural experiments. The updated classes and attributes of the Lehigh Model have improved the representation of this information and data. The Lehigh RTMD Metadata Web was developed to implement the Lehigh Model. The two examples given in this thesis show how the Lehigh RTMD Metadata Web can be used to organize the information and data related to structural experiments. In addition, the examples show how users can access and retrieve the information and data.

#### 7.3 **Recommendations for Future Work**

Future work on the Lehigh Model should focus on developing classes and attributes for other large-scale structural test methods, such as Tsunami wave tank experiments, shaking table experiments, and centrifuge experiments.

Future work on the Lehigh RTMD Metadata Web should focus on updating the web application as the Lehigh Model is improved, and on developing a search tool to help users to quickly find specific information and data.

# References

Demitzakis, S. N. and Mahin, S. A. (1985). "Development of Substructuring Techniques for On-Line Computer Controlled Seismic Performance Testing," Report UBC/EERC-85/04, Earthquake Engineering Research Center, University of California, Berkeley CA.

Hong, N. K. and Sause, R. (1994). "Concepts and Notation for Integrated Structural Design: Product and Process Models," ATLSS Report No. 94-13, Advanced Technology for Large Structural Systems (ATLSS) Center, Lehigh University, Bethlehem PA.

http://nisee.berkeley.edu/spd/, web site for PEER Structural Performance Database, Pacific Earthquake Engineering Research Center, Berkeley, CA.

http://www.sacsteel.org/, SAC Steel Project web site, SAC Steel Project, Richmond CA.

Lee, C., Chin, C. H., Marullo, T., Bryan, P., Sause, R., and Ricles, J. M. (2006). "Development of Data Model for Large-scale Structural Experiments." Advanced Technology for Large Structural Systems (ATLSS) Center, Lehigh University, Bethlehem PA.

Marullo, T. (2007) "Lehigh RTMD Metadata Web." ATLSS Center, Lehigh University, Bethlehem, PA. (available at http://www.nees.lehigh.edu/metadataweb/).

Mercan, O. (2007). "Analytical and Experimental Studies on Large Scale, Real-time Pseudodynamic Testing," Ph.D. Dissertation, Department of Civil and Environmental Engineering, Lehigh University, Bethlehem, PA.

NEESit (2006). "NEEScentral 1.5 User's Guide," George E. Brown, Jr. Network for Earthquake Engineering Simulation (NEES), NEES Cyberinfrastructure Center (NEESit), University of California, San Diego, La Jolla CA, (available at http://it.nees.org/).

Oregon State University (2003). "NEES Database and Metadata Structure, Version 1.3," white paper, by Oregon State University and Network Alliance for Computational Science and Engineering, Network for Earthquake Engineering Simulation, (available at http://nees.orst.edu/IT/data.model/docs/v1.3a.july16.pdf).

Peng, J., and Law, K. (2004). "Reference NEESgrid Data Model," Technical Report NEESgrid-2004-40, Network for Earthquake Engineering Simulation, (available at http://it.nees.org/documentation/pdf/TR-2004-40.pdf).

Ricles, J. M., Sause, R., Garlock, M. M., and Zhao, C. (2001). "Posttensioned Seismic-Resistant Connections for Steel Frames," *Journal of Structural Engineering*, ASCE, Vol. 127, No. 2, 113-121.

Wolski, M.E. (2006). "Experimental Evaluation of a Bottom Friction Device for a Self Centering Seismic Moment Resistant Frame with Post-Tensioned Steel Moment Connections," M.S. Thesis, Department of Civil and Environmental Engineering, Lehigh University, Bethlehem, PA.

## Vita

Stanley C.H. King was born on November 4, 1979 in Taipei, Taiwan, R.O.C.. He is son of Nan Chan Ching and Li-Wa King-Huang. Stanley received his Bachelor of Science degree in Civil Engineering at Tamkang University, Tamsui, Taiwan, R.O.C. in June 2002. He then went to U.S to pursue his Master of Science degree in Civil Engineering at Lehigh University, Bethlehem, Pensonsylvania. He will receive his degree in May 2011. He is working as a structural engineer in Goldstein Associations in New York City in the past 3 years and will be continuously working over there.